ISSN 0103-4065



Jornal Brasileiro de Doenças Sexualmente Transmissíveis

**BJSTD** publishing new scientific knowledge.

Official Organ of the Brazilian Society for Sexually Transmitted Diseases Official Organ of the Latin American and Caribbean for Control of STDs Official Organ for Latin America Union Against International Sexually Transmitted Infections Official Organ of the Portuguese Papillomavirus Society Official Organ of the Sexually Transmitted Diseases Sector - Fluminense Federal University



VOLUME 31 Nº 2 2019

# **CONTENTS**

### **EDITORIAL**

WHY IS THE NONAVALENT HPV VACCINE SO IMPORTANT FOR BRAZIL?	39
POR QUE A VACINA NONAVALENTE CONTRA O HPV É TÃO IMPORTANTE PARA O BRASIL?	
Edison Natal Fedrizzi	
EDITORS' NOTE ON NONAVALENT HPV VACCINE (9HPV)	43
NOTA DOS EDITORES SOBRE A VACINA NONAVALENTE CONTRA O HPV (9HPV)	
Mauro Romero Leal Passos, José Eleutério Junior	

### ARTICLES

### **REVIEW ARTICLE**

### **CASE REPORTS**

### **OPINION**

### EVENTS



OFFICIAL ORGAN OF THE BRAZILIAN SOCIETY FOR SEXUALLY TRANSMITTED DISEASES

Av. Roberto Silveira, 123 - Niterói - RJ - Brazil CEP: 24230-150 - Tel.: + 55 (21) 2710-1549 www.dstbrasil.org.br

SBDST BOARD (2017-19)

President: Mauro Romero Leal de Passos (RJ)

Vice-President: Paulo César Giraldo (SP)

1º Secretary: José Eleutério Junior (CE)

**2º Secretary:** Isabel Cristina C. do Val Guimarães (RJ)

1º Treasurer: Renato de Souza Bravo (RJ)

**2º Treasurer:** Edilbert Nahn Pellegrini Junior (RJ)

Scientific Director: Angelica Espinosa Miranda (ES)

Supervisory Board: Ivo Castelo Branco Coêlho (CE) Maria Luiza Bezerra Menezes (PE) Newton Sergio de Carvalho (PR)

REGIONAL SBDST: REGIONAL ESPÍRITO SANTO President: Sandra Fagundes Moreira Silva

REGIONAL PARANÁ President: Ali Bakri

REGIONAL PERNAMBUCO President: Maria Carolina Pessoa Valença Rygaard

REGIONAL RIO DE JANEIRO President: Mauro Romero Leal Passos

REGIONAL RIO GRANDE DO SUL President: Mauro Cunha Ramos

REGIONAL SÃO PAULO President: Elucir Gir

Access the app of the journal **Sexually Transmitted Diseases** 

for free in Smartphones and Tablets (iOS and Android)

Download the Apps and have access to the content at the touch of a hand.



OFFICIAL ORGAN OF THE LATIN AMERICAN AND CARIBBEAN FOR CONTROL OF STDS



DST - BJDST is the official organ for Latin America Union Against International Sexually Transmitted Infections (IUSTI) Regioal Representative: Adele Benzaken



Sociedade Portuguesa de Papilomavírus Presidente: Rui Medeiros Secretária-Geral: Maria Clara Bicho



### **EDITORIAL COUNCIL**

Editir-in-chief Mauro Romero Leal Passos

Editors: Angelica Espinosa Miranda (ES) Newton Sérgio de Carvalho (PR) José Eleutério Junior (CE) Mariângela Silveira (RJ) Paulo César Giraldo (SP)

Assistant Editors: Dennis de Carvalho Ferreira (RJ) Edilbert Nahn Pellegrini Junior (RJ) Felipe Dinau Leal Passos (RJ) Renata de Queiroz Varella (RJ)

### **Editorial Board:**

Adele Schwartz Benzaken (AM) Felipe Piedade Gonçalves Neves (RJ) Geraldo Duarte (SP) Gutemberg Leão de Almeida Filho (RJ) Helena Rodrigues Lopes (RJ) Iara Moreno Linhares (SP) Isabel Cristina C.V. Guimarães (RJ) Ivo Castelo Branco Coêlho (CE) Izabel Christina NP Paixão (RJ) Maria Luiza Bezerra Menezes (PE) Marcia Soares Pinheiro (RJ) Otílio Machado Pereira Bastos (RJ) Roberto de Souza Salles (RJ) Rosane Figueiredo Alves (GO) Silvia Maria B Cavalcanti (RJ) Vandira Maria dos Santos Pinheiro (RJ)

### International Editorial Board:

Alícia Farinati (Argentina) Enrique Galbán García (Cuba) Maria Clara Bicho (Portugal) Steven Witkin (EUA)

**Edition Assistant:** Rubern de Avelar Goulart Filho (RJ)

Secretary: Dayse Felício (RJ)

Publication and Copydesk: Zeppelini Publishers / Rede Filantropia www.zeppelini.com.br / www.filantropia.ong OFFICIAL ORGAN OF THE SECTOR SEXUALLY TRANSMITTED DISEASES

MINISTÉRIO DA EDUCAÇÃO UNIVERSIDADE FEDERAL FLUMINENSE SETOR DE DOENÇAS SEXUALMENTE TRANSMISSÍVEIS

> Outeiro de S. João Batista, s/n<sup>2</sup> Campus do Valonguinho - Centro Niterói - RJ - 24210-150 - Brazil Tel.: +55 (21) 2629-2495 - 2629-2494 Fax: +55 (21) 2629-2507

> > E-mail: dst@dst.uff.br http://www.bjstd.org/

Rector of UFF: Antonio Claudio Lucas da Nóbrega Vice-Rector: Fabio Barboza Passos Provost of Research, Post-Graduate and Innovation: Andrea Brito Latge

> **Biomedical Institute:** Otílio Machado Pereira Bastos

Department of Microbiology and Parasitology: Claudia Maria AU Souto Maior

> Chief of STD Sector: Mauro Romero Leal Passos



Órgão de divulgação do Programa de Pós-graduação Mestrado e Doutorado em Microbiologia e Parasitologia Aplicadas, Mestrado profissional em Saúde Matero-Infantil, ambos da UFF e de outros da área médica e Ciências da Saúde.

### Editora da UFF Editora da Universidade



ditora da Universidade Federal Fluminense

Matters signed and published in DST - Brazilian Journal of Sexually Transmitted Diseases are solely the responsibility of their respective authors and do not necessarily reflect the opinion of the editors.

### **Targeting and Distribution:**

DST - Brazilian Journal of Sexually Transmitted Diseases is directed to members of SBDST, subscribers, libraries, reference centers, gynecologists, urologists, infectious disease specialists, dermatologists, clinicians, family health programs and entities with an agreement. It is quarterly with a circulation of 3,000 copies.

> Exchange requested - Pode-se permuta On prie l'échange - Se solicita ei caxzje Mau bitet nu Austausch - Si prega lo escambo

All content of the whole collection since 1989 is available for free on the World Wide Web at www.dst.uff.br

### INDEXING:

LILACS EXPRESS Latin American Literature in Health Sciences, The Library of Congress WC – 140, Scholar google

Since the references are given in full (name of the article, all authors names, journal name, year, volume, page numbers and the site: www.dst.uff.br), the reproduction is allowed in whole or in part, just one copy of this journal, for personal use only, never for commercial purposes. EDITORIAL

# WHY IS THE NONAVALENT HPV VACCINE SO IMPORTANT FOR BRAZIL?

### Por que a vacina nonavalente contra o HPV é tão importante para o Brasil?

Edison Natal Fedrizzi<sup>1</sup>

### ABSTRACT

Genital Human papillomavirus (HPV) infection is the most common sexually transmitted infection worldwide. It presents from latent infection to anogenital and oropharyngeal carcinoma in both men and women. Cervical cancer is still the most prevalent cancer associated with HPV and in many countries, such as Brazil, is considered an epidemic, although Pap smears are available in public and considered a way of prevention of the disease. However, in Brazil, the coverage of this test is very low, reflecting the maintenance of high mortality for this disease for decades. The most effective way to prevent this disease, of course, is to prevent HPV infection with vaccination. High vaccine coverage using the 4HPV vaccine, which is available in Brazil, would provide a significant reduction in cancer in our country, whereas the use of 9HPV would prevent almost all high-grade lesions, true precursors of cervical cancer. **Keywords:** Papillomavirus infections; Papanicolaou test; papillomavirus vaccines.

### RESUMO

A infecção genital pelo papilomavírus humano (HPV) é a infecção de transmissão sexual mais frequente no mundo inteiro. Ela se apresenta desde a infecção latente até o carcinoma da área anogenital e orofaringe em homens e mulheres. O câncer de colo de útero ainda é o mais prevalente associado ao HPV e em muitos países, como no Brasil, é considerado uma epidemia, apesar de o exame de Papanicolaou estar disponível nas redes públicas e ser considerado uma forma de prevenção da doença. No entanto, no Brasil, a cobertura deste exame é muito baixa, refletindo na manutenção da alta mortalidade por esta doença há décadas. A forma mais eficaz de prevenção desta doença, sem dúvida, é a prevenção da infecção pelo HPV através das vacinas. Uma alta cobertura vacinal, utilizando a vacina 4HPV, que é a disponível no Brasil, proporcionaria uma redução importante do câncer em nosso meio, ao passo que a utilização da 9HPV, preveniria a quase a totalidade das lesões de alto grau, verdadeiras precursoras do câncer de colo de útero. **Palavras-chave:** Papillomaviridae; infecções por papillomavirus; teste de Papanicolau; vacinas contra papillomavirus.

### INTRODUCTION

Genital HPV infection is the most frequent Sexually Transmitted Infection (STI) in both women and men<sup>(1)</sup>. It is estimated that at least 50% of sexually active individuals will enter in contact with HPV at some point in their lives, and that 80% of women will have this contact up to 50 years of age<sup>(2)</sup>. We can consider this an epidemic infection in Brazil, because there are from 9 to 10 million infected people with this virus, and 700,000 more new cases are found every year<sup>(3)</sup>.

Recently, preliminary results of the prevalence of HPV infection in Brazil were published in the POP Study (Papillomavirus Prevalence Study in Brazil), a cross-sectional study with 7,586 individuals sexually active (5,812 women and 1,774 men) from 16 to 25 years old, in the 26 capitals of Brazil and the Federal District. HPV prevalence was 54.6% with positive high-risk HPV in 38.4% of the participants. The highest prevalence was in Salvador City, with a positivity rate of 71.9%, and the lowest rate in Recife, with 41.2%<sup>(4)</sup>.

Worldwide, 32 million new cases of genital warts are estimated each year (Brazil, around 1.9 million/year), and the vast majority is associated to HPV 6 (70% of cases) and 11 (20% of cases)<sup>(5)</sup>.

### **HPV INFECTION AND CANCER**

For the biennium 2018-2019, 16,370 new cases of cervical cancer per year in Brazil are estimated, with a risk of 15.43 cases/100,000 women occupying the third position, only after breast and colon/rectum cancer. However, without considering non-melanoma skin cancer, cervical cancer is the first most incident in the Northern region (25.62/100,000 women)<sup>(6)</sup>. The estimate of deaths is 8,079 cases/year (7.5/100,000 women)<sup>(7)</sup>. If we consider other types of HPV-related cancers, we can also consider HPV-induced cancer an epidemic in Brazil (**Table 1**)<sup>(7)</sup>.

HPV types 16, 18, 31, 33, 35, 39, 45, 51, 52, 56, 58, 59, 68, 82, 26, 53, 66 and 73 are the most often considered high oncogenic risk, and HPV types 6, 11, 40, 42, 43, 44, 53, 54, 54, 61, 70, 72, 73, 81, CP6108, low risk<sup>(8)</sup>. In general, HPV 16 and 18 account for 70% of cases of cervical cancer, and 80-90% of cases of HPV cancers induced in other areas, whereas HPV 6 and 11 are the causes of at least 90% of anogenital warts and almost all larynx papillomatoses<sup>(9)</sup>. High-risk HPV is still responsible for cancer of other organs, such as anus, in 90%; 70% of the vagina, 50% of the penis; 40% of the vulva; and 13-72% of the oropharynx<sup>(10)</sup>.

### **HPV VACCINE EFFICACY**

Currently, the most effective way to prevent infection and HPVassociated diseases is the use of vaccines. Up to date, three vaccines are available on the international market: the bivalent or Cervarix<sup>®</sup> (2HPV — protection against HPV 16 and 18), the quadrivalent or

Table 1 –	Crude incidence rates of HPV-related cancers/	/100,	000.
-----------	---	-------	------

	Male	Female
Cervical cancer	-	15.2
Anal cancer	0.3-1.2	0.8-1.7
Vulvar cancer	-	0.8-2.7
Vaginal cancer	-	0.0-0.9
Penile cancer	0.3-2.0	-
Oropharyngeal cancer	3.7	0.8

HPV: Human papillomavirus.

Source: ICO/IARC Information Centre. Human Papillomavirus and Related Diseases Report – Brazil, 2019<sup>(7)</sup>.

<sup>&</sup>lt;sup>1</sup>Universidade Federal de Santa Catarina – Florianópolis (SC), Brazil.



Figure 1 - Comparison of the ten most frequent human papillovirus (HPV) oncogenic types in Brazil among women with and without cervical lesions.

**Table 2 –** Human papillovirus (HPV) types in the cervix with and without lesion in Brazil.

HPV INFECTION	HPV 16, 18 (%)	HPV 16, 18, 31, 33, 45, 52, 58 (%)
Normal Cytology	5.4	9.2
LSIL	30.8	62.3
HSIL	56.8	99.6
Cervical cancer	68.2	86.5

LSIL: low-grade lesion; HSIL: high-grade lesion.

Source: ICO/IARC Information Centre. Human Papillomavirus and Related Diseases Report – Brazil, 2019<sup>(7)</sup>.

Gardasil<sup>®</sup> (4HPV — against HPV 6, 11, 16, and 18), and the nonavalent or Gardasil 9<sup>®</sup> (9HPV — against HPV 6, 11, 16, 18, 31, 33, 45, 52 and 58), which were highly effective and safe in clinical trials with rates of 95-100% efficacy in the prevention of genital warts and precancerous lesions of the lower genital tract associated to HPV 16 and 18 (2HPV and 4HPV) and for the other five types of highrisk HPV for nonavalent vaccine<sup>(11)</sup>. Although the 9HPV vaccine has been approved by the Brazilian National Health Surveillance Agency (ANVISA) in December 2017, it is not available in the national market yet.



Figure 2 – Comparison of specific mortality coefficient by cervix cancer among the five Brazilian regions and Brazil over nine years (2006 to 2014) per 100,000 women.

The most appropriate moment for the use of HPV vaccine is before exposure to the virus. However, the most recent studies also show benefits for already infected women, including those with cervical intraepithelial neoplasia (CIN) 2/3 lesions, showing a decrease in recurrences of roughly 75–88% for vaccinated women<sup>(12)</sup>. Although there is an age recommendation for the use of the vaccines in the label, they were highly safe, immunogenic, and effective also in elderly men and women<sup>(12)</sup>.

# HPV DISEASES AND HPV VACCINATION IN BRAZIL

Since March 2014, the 4HPV vaccine is available on the public vaccination calendar in Brazil. Currently, the National Immunization Program (PNI) provides the vaccine to girls aged from 9 to 14, and boys aged from 11 to 14, as well as men and women aged from 9 to 26 years with an immune problem (HIV infection, radio/chemotherapy for cancer, bone marrow and solid organs transplantation<sup>(13,14)</sup>.

When we evaluate the Brazilian scenario as to HPV types among women with and without cervical lesion (Figure 1), we observe a great variability between the different types of HPV and the severity of lesions<sup>(7)</sup>.

### THE IMPORTANCE OF 9HPV VACCINE

When we classify HPV lesions associated to viruses 16 and 18 (direct protection with the 2HPV and 4HPV vaccine), and viruses 16, 18, 31, 33, 45, 52, and 58 (direct protection with the 9HPV vaccine), we observed a significant increase in the protection of lesions with the 9HPV vaccine. If we do not consider the possible cross-protection and consider a protection of 100% of the vaccine, we will be protecting a low-grade lesion (LSIL) by 30.8% using 2HPV and 4HPV vaccines, whereas we double this protection to 62.3% when we use the 9HPV vaccine. This difference becomes more evident for real precursor lesions of cancer, high-grade lesions (HSIL), where we obtain a protection of 56.8% with 2HPV and 4HPV vaccines, and to virtually, the totality of the cases (99.6%) with the vaccine 9HPV<sup>(7)</sup> (**Table 2**).

Unfortunately, the mortality by cervical cancer remained unchanged in several decades in Brazil, regardless of the various programs and campaigns carried out over these years for the Pap smear use, as demonstrated in a recent analysis of nearly 10 years in our country (**Figure 2**)<sup>(15)</sup>. It is quite clear that the most effective way to prevent this disease is with the HPV vaccination. Brazilian health authorities must take urgent actions to get high HPV vaccination coverage again (as occurred with the first dose in the year of implementation of the public vaccination program) by restarting vaccination in schools, spreading information for the public and health professionals, facilitating the access to Basic Health Units (expanding working hours), and adopting the 9HPV vaccine as soon as possible, so that we can definitively change this reality in Brazil, of about 16,000 new cases/year and 8,000 deaths/year by cervical cancer, a disease absolutely preventable.

### CONCLUSION

Cervical cancer mortality in Brazil has remained the same for several decades, regardless of public campaigns and programs to encourage it, especially due to its low coverage. One of the best strategies to change this scenario is HPV vaccination, which should achieve high vaccination coverage and replacement of 4HPV vaccine for 9HPV as soon as possible.

### Funding

The author declares there are no grants or other funding for all authors.

### **Conflict of interests**

Researcher of the quadrivalent and nonavalent vaccine trials in men and women by the company MSD.

### **Ethics Committee**

There is no need for this action.

### REFERENCES

- Burchell NA, Winer RL, Sanjosé S, Franco EL. Chapter 6: Epidemiology and transmission dynamics of genital HPV infection. Vaccine. 2006;24(Suppl. 3):S3/52-61. https://doi.org/10.1016/j.vaccine.2006.05.031
- Castellsagué X, Bosch FX, Muñoz N. Environmental co-factors in HPV carcinogenesis. Virus Res. 2002;89(2):191-9. https://doi.org/10.1016/ s0168-1702(02)00188-0
- Giraldo PC, Silva MJPMA, Fedrizzi EM, Gonçalves AKS, Amaral RLG, Eleutério Jr. J, et al. Prevenção da infecção por HPV e lesões associadas com o uso de vacinas. J Bras Doenças Sex Transm. 2008;20(2):132-40.
- Associação Hospitalar Moinhos de Vento. Estudo epidemiológico sobre a prevalência nacional de infecção pelo HPV (POP-Brasil): Resultados preliminares. Porto Alegre: Associação Hospitalar Moinhos de Vento; 2017.
- Sadjadi A, Malekzadeh R, Derakhshan MH, Sepehr A, Nouraie M, Sotoudeh M, et al. Cancer occurrence in Ardabil: results of a populationbased cancer registry from Iran. Int J Cancer. 2003;107(1):113-8. https:// doi.org/10.1002/ijc.11359
- Instituto Nacional de Câncer José Alencar Gomes da Silva. Coordenação de Prevenção e Vigilância. Estimativa 2018: incidência de câncer no Brasil. Rio de Janeiro: INCA; 2017.
- ICO/IARC HPV Information Centre. Human Papillomavirus and Related Diseases Report – Brazil. ICO/IARC HPV Information Centre; 2019.
- Baseman JG, Koutsky LA. The epidemiology of Human papillomavirus infections. J Clin Virol. 2005;32(Suppl. 1):S16-S24. https://doi. org/10.1016/j.jcv.2004.12.008
- Muñoz N, Castellsagué X, de González AB, Gissmann L. HPV in the etiology of human cancer. Vaccine. 2006;24(Suppl. 3):S3/1-10. https:// doi.org/10.1016/j.vaccine.2006.05.115
- Ferlay J, Shin HR, Bray F, Forman D, Mathers C, Parkin DM. Estimates of worldwide burden of cancer in 2008: GLOBOCAN 2008. Int J Cancer. 2010;127(12):2893-917. https://doi.org/10.1002/ijc.25516
- Arbyn M, Xu L, Simoens C, Martin-Hirsch PPL. Prophylactic vaccination against Human Papillomaviruses to prevent cervical cancer and its precursors (Review). Cochrane Database Syst Rev. 2018;5:CD009069. https://doi.org/10.1002/14651858.CD009069.pub3
- Kang WD, Choi HS, Kim SM. Is vaccination with quadrivalent HPV vaccine after loop electrosurgical excision procedure effective in preventing recurrence in patients with high-grade cervical intraepithelial neoplasia (CIN 2/3)? Gynecol Oncol. 2013;130(2):264-8. https://doi.org/10.1016/j.ygyno.2013.04.050
- 13. Brasil. Ministério da Saúde. Secretaria de Vigilância em Saúde. Departamento de Vigilância das Doenças Transmissíveis. Coordenação Geral do Programa Nacional de Imunizações. Nota Informativa referente às mudanças no Calendário Nacional de Vacinação para o ano de 2017. Brasília: Ministério da Saúde; 2017.
- Brasil. Ministério da Saúde. Coordenação Geral do Programa Nacional de Imunizações. Informe Técnico das vacinas papilomavírus humano 6, 11, 16 e 18 (recombinante) – vacina HPV quadrivalente e meningocócica C conjugada. Brasília: Ministério da Saúde; 2018.
- Fedrizzi EN, Ponce NM. Coverage of Pap smear and mortality from cervical cancer in Brazil from 2006 to 2014. J Bras Doenças Sex Transm. 2017;29(4):117-24. http://doi.

# **EDITORS' NOTE ON NONAVALENT HPV VACCINE (9HPV)**

### Nota dos editores sobre a vacina nonavalente contra o HPV (9HPV)

Mauro Romero Leal Passos<sup>1</sup> , José Eleutério Junior<sup>2</sup>

In the first place, I would like to emphasize and thank the great initiative and exceptional Editorial by Professor Edson Fedrizzi.

Despite all that has been said, we should not fail to mention that the frequent use of male or female condoms (long forgotten in educational campaigns, of free distribution in public events and put for sale at commercial establishments) is an essential method to prevent all sexually transmitted diseases (STDs), including the Human papillomavirus HPV, hepatitis B, and HIV. Frequently, people infected with STD are those who already acquired another STD. Vaccination against HPV does not protect these people against other STDs. Therefore, since there are no vaccines against classic STDs (except for hepatitis B), any information on HPV should (and, in fact, have) support vast actions against all other STDs.

On the other hand, appeals for the prevention of HPV infections and associated diseases should not insist on exclusive statements that this is a vaccine against cervical cancer. In our point of view, keeping on spreading messages that cause fear/terror with "heavy" words, such as cancer, means missing a valuable opportunity to create a "kind" channel to broadly talk about sex and STDs, without prejudice, to the entire population, especially adolescents. We point out that this is also the vaccine against premalignant and malignant lesions of the penis, anus, vulva and oropharynx.

Sure of the importance of the HPV vaccine, we support the expansion of the administration by the Brazilian public health system for people of both genders, aged from 9 to 17, *people of both genders up to 45 years of age* living with HIV, solid organs transplanted people, bone marrow, oncologic, people victims of rape and sexual abuse, and for users of pre- and post-HIV prophylaxis (PrEP and PEP). These groups of people have the most severe outcomes when affected by HPV. At this moment, 9HPV vaccine is the best way to prevent HPV infections/diseases.

We must also get out of the monotony that is up to the public sector only to fight STDs. All sections and actors, including health and education professionals (public, private, and non-governmental organizations), public and private health managers/supplementary medicine (diseases prevention and treatments impact corporate finances), professionals from health scientific societies and education, communication and social sciences, media professionals in general, social influencers, and people in general are also responsible, considering we lack books, plays, movies, social media posts, lectures, seminars, articles, and interviews in newspapers and magazines on the matter, that is, general pieces of raising awareness. Every action leads to more information, dialogues about sexuality, and emotional and social repercussions involving STDs. It will provide the basis for correct information, minimizing the impact of fake news and actions of anti-vaccine groups<sup>(1)</sup>.

For a long time, since the pre-launch, launch and post-launch dissemination of bi and quadrivalent vaccines against HPV in Brazil, we have been "flooded" by vaccine producing companies with the speech that we have to make and offer what is most effective against infections/diseases caused by different types of HPV. Thus, we work on a day-to-day basis on various problems.

Today, the world made available a vaccine with a higher action spectrum against HPV. However, as already said, the 9HPV vaccine has been approved in Brazil since 2017. However, it is not available neither to the Brazilian population (in the public and private sphere) nor to several other countries in Latin America and Africa. Areas in which infections/diseases (including cancer) caused by HPV are the largest in the world. Please refer to the Brazilian data already available.

We have information that the 9HPV vaccine is already available in China, where the public service, as far as we know, has not sponsored the purchase of 4HPV vaccine for public vaccination in same the scale that Brazil has implemented and implements since 2014.

As informed by immunization market professionals, the producer of 4HPV and 9HPV vaccines is having trouble to meet the demands, especially with the entry in the Chinese market.

The commercialization of the 9HPV vaccine to new countries, however, given that the manufacturer cannot produce for all the market, made impossible to the Brazilian population (public and private) to have the best product in the prevention/ fight against infections/diseases caused by HPV, because this is an evolution of 4HPV, implemented in Brazil in the public network since 2014, and commercialized, concurrently, in the private system since 2007.

Situations like these strengthen the speech that above respecting and solidifying advances in public health, by the concept of private pharmaceutical companies, is the opening of new fields of business, trades and profit. Issues such as improving people's health are secondary<sup>(2)</sup>.

As far as we know, the 9HPV vaccine is available in public and private spheres in the United States, Canada, Chile, Ecuador, Peru, Croatia, Denmark, Scotland, Gibraltar, Italy, Lietchtentein, Wales, Portugal, United Kingdom, Czech Republic, Serbia, Switzerland, Turkmenistan, Israel, Qatar, South Korea, Singapore, Australia, New Zealand, China, Hong Kong<sup>(3)</sup>.

When will Brazil have the 9HPV vaccine available, at least for the private system, like what happened with 2HPV and 4HPV?

<sup>&</sup>lt;sup>1</sup>Setor de Doenças Sexualmente Transmissíveis, Universidade Federal Fluminense – Niterói (RJ), Brazil.

<sup>&</sup>lt;sup>2</sup>Faculdade de Medicina, Universidade Federal do Ceará – Fortaleza (CE), Brazil.

### Participation of each author

The authors declare that all authors were active participants.

### Funding

The authors declare there are no grants or other funding for all authors.

### **Conflict of interests**

The is no conflict of interest to declare.

### **Ethics Committee**

There is no need for this action.

### REFERENCES

- Brasil. Câmara dos Deputados. A eficácia e a faixa etária da administração da vacina HPV: Audiência Pública [Internet]. Brasil: Câmara dos Deputados; 2019 [cited on Oct. 30, 2019]. Available at: https://edemocracia.camara.leg.br/audiencias/sala/1305
- Angell M. A verdade sobre os laboratórios farmacêuticos. Rio de Janeiro: Record; 2007.
- Giuliano A. HPV In Males: Rational For Gender Neutral Vaccination. In: 37<sup>th</sup> Annual Meeting of the European Society for Pediatrics Infectious Diseases; 2019, Ljubljana, Slovenia. Annals. 2019.

### Address for correspondence: *MAURO ROMERO LEAL PASSOS* Rua Amapá, 22, ap. 503 – São Francisco

Niterói (RJ), Brazil CEP: 24365-100. E-mail: maurodst@gmail.com

# ANALYSIS OF MONITORED CASES OF ACQUIRED SYPHILIS IN PORTO ALEGRE, RIO GRANDE DO SUL

### Análise de casos monitorados de sífilis adquirida em Porto Alegre, Rio Grande do Sul

Gustavo Haas Lermen<sup>1</sup>, Lisiane Morelia Weide Acosta<sup>2</sup>, Fabiana Ferreira dos Santos<sup>3</sup>, Juanita Natasha Garcia de Oliveira<sup>1</sup>

### ABSTRACT

**Introduction:** Syphilis is a systemic and curable sexually transmitted infection, exclusive to humans. Brazil is experiencing an increasing incidence of syphilis in recent years. Porto Alegre has an incidence rate of acquired syphilis (AS) four times higher than the national one. **Objective:** To describe the epidemiological profile of monitored cases of AS, through rapid tests, in the city of Porto Alegre/Rio Grande do Sul in the first semester of 2018. **Methods:** The study has a cross-sectional, observational, epidemiological, and analytical design, and the study population was included in the Porto Alegre AS monitoring spreadsheet in the first semester of 2018, totaling 1,453 participants. We performed descriptive analyses with absolute and relative frequencies to develop the profile. The  $\chi^2$  test for bivariate analysis verified the association between sociodemographic variables and the adequate treatment outcome, with a 95% significance level. **Results:** The profile found had a predominance of males, white people, individuals aged 20 to 29, with incomplete elementary school, and not homeless. We identified an association in the bivariate analysis between the adequate treatment outcome and the variables schooling, gender, age, and homelessness. **Conclusion:** Actions that seek to structure and organize the processes related to AS monitoring are important, especially due to the association of sociodemographic variables that indicate social vulnerability with the adequate treatment outcome. **Keywords:** syphilis; epidemiological monitoring; epidemiology; monitoring.

RESUMO

**Introdução:** A sífilis é uma Infecção Sexualmente Transmissível de caráter sistêmico, curável, e exclusiva do ser humano. O Brasil vive um período de crescimento dos casos de sífilis nos últimos anos. Porto Alegre possui um coeficiente de incidência de sífilis adquirida quatro vezes maior que o nacional. **Objetivo:** O objetivo deste estudo foi descrever o perfil epidemiológico dos casos monitorados para sífilis adquirida, por meio de testes rápidos, no município de Porto Alegre/RS, no primeiro semestre de 2018. **Métodos:** O delineamento é epidemiológico observacional analítico transversal, sendo todos os casos inseridos na planilha de monitoramento da SA de Porto Alegre no primeiro semestre de 2018, que totalizou 1.453 usuários. Análises descritivas foram realizadas com frequência absoluta e relativa para caracterizar o perfil. Foi utilizado o teste de  $\chi^2$  em análise bivariada para verificar a associação das variáveis sociodemográficas com o desfecho do tratamento adequado, com nível de significância de 95%. **Resultados:** O perfil encontrado foi predominância do sexo masculino, brancos, na faixa etária de 20 a 29, ensino fundamental incompleto e não estar em situação de rua. Houve associação na análise bivariada entre os desfechos do tratamento adequado e as variáveis escolaridade, sexo, faixa etária e situação de rua. **Conclusão:** Ações que buscam estruturar e organizar os processos relativos ao monitoramento da SA são importantes, especialmente por haver associação de variáveis sociodemográficas que indicam vulnerabilidade social com o desfecho de tratamento adequado da sífilis. **Palavras-chave:** Sífilis; Vigilância Epidemiológica; Epidemiologia; Monitoramento.

### INTRODUCTION

Brazil has been experiencing a growing incidence of syphilis in recent years. In 2016, 87,593 cases of acquired syphilis (AS) were reported. Stratified by regions, the Southern Region holds the second place, corresponding to 24.2% of the reported cases. In addition, the detection rate in Brazil in the aforementioned year was 42.5 cases of AS per 100,000 inhabitants. The South Region exceeded this rate with 72 cases per 100,000 inhabitants, while Rio Grande do Sul (RS) had the highest rate in the country (93.7 cases per 100,000 inhabitants)<sup>(1)</sup>. The capital of the state, Porto Alegre (POA), had a coefficient of AS incidence four times greater than the national one<sup>(2)</sup> and was the fifth capital with the highest detection rate according to data from the General Coordination of Health Surveillance (*Coordenadoria Geral de Vigilância em Saúde* – CGVS)<sup>(3)</sup>.

Syphilis is a systemic and curable sexually transmitted infection (STI), exclusive to human beings, caused by the bacterium *Treponema pallidum*. Estimates indicate that 12 million new cases of syphilis occur annually in the world adult population<sup>(4)</sup>. The infection is transmitted by sexual contact, direct contact with open lesions, transfusion of contaminated blood, and transplacentally<sup>(5)</sup>.

Most people with syphilis tend not to be aware of the pathology, due to the absence of symptoms, depending on the stage of infection, which can enable transmission to their sexual partners. Thus, it is a silent disease getting worse over the years, reaching other stages, and, despite having effective diagnosis and treatment, which is free for the population, it remains a public health problem until today.

For case detection, the general population of Porto Alegre has access to free rapid tests (RT), without following the regionalization logic, that is, the user has the right to undergo the procedure in any Basic Health Unit (BHU) without the need for scheduling<sup>(6)</sup>. All pregnant women should have an RT in the first and third trimester of pregnancy, according to the low-risk prenatal care protocol<sup>(7)</sup>.

The increasing identification and diagnosis with the advent of the RT stand out, since this tool is fundamental for investigation and

<sup>&</sup>lt;sup>1</sup>Specialization in Public Health, Universidade Federal do Rio Grande do Sul – Porto Alegre (RS), Brazil.

<sup>&</sup>lt;sup>2</sup>Ph.D. in Collective Health, Universidade do Vale do Rio dos Sinos – Porto Alegre (RS), Brazil.

<sup>&</sup>lt;sup>3</sup>Master in Health Sciences, Universidade Federal do Rio Grande do Sul – Porto Alegre (RS), Brazil.

necessary referrals, aiming at comprehensive, effective, and successful care<sup>(5)</sup>. However, there are gaps regarding the systematization of the follow-up required for the confirmation of diagnosis, treatment, outcome, and, consequently, the reduction in the *Treponema* transmission chain.

The Research Ethics Committee of the Municipal Health Department of Porto Alegre approved this study under protocol no. 2931412.

In order to transpose the test to address the high incidence problem, the Municipal Health Department (*Secretaria Municipal de Saúde* — SMS) began a monitoring process of rapid tests reactive for AS in BHUs of the Primary Care Network (*Rede de Atenção Primária* — RAP) and Specialized Care Services (*Serviços de Atenção Especializada* — SAE) of the city of Porto Alegre in the state of Rio Grande do Sul.

This article aims at analyzing this work and its initial results. Therefore, to evaluate the situation of the city and its regions, the analysis was also carried out by District Administrations (DAs), administrative structures, and regional managers, which are the agencies responsible for discussion and practice, where all health care strategies of the public health system are elaborated. DAs are distributed in eight health regions, with defined geographical boundaries and population to determine the services offered<sup>(8)</sup>.

### **OBJECTIVE**

To describe the epidemiological profile of monitored AS cases based on RTs in the city of Porto Alegre/Rio Grande do Sul in the first semester of 2018.

### **METHODS**

This is an analytical cross-sectional epidemiological study. The study population comprised all cases entered in the Porto Alegre AS monitoring spreadsheet in the first semester of 2018, excluding pregnant women or those who became pregnant before the completion of the treatment, and users who refused any form of treatment.

The AS monitoring spreadsheet only has the cases of RT reactive for syphilis, deriving from the electronic form completed by health services with information regarding the tests in general, shared between care and management, generating tables for different DA services for each corresponding month.

Data from the AS monitoring spreadsheet corresponding to the first semester of 2018 were compiled and stored using Google Drive Sheets, and included the independent sociodemographic variables of the survey: the BHU that performed the test and its respective DA, gender, ethnicity, age, schooling, housing situation, and date of completion of the RT. We emphasize that sociodemographic factors were self-reported.

In the monitoring process, the evaluation of the pathology treatment consisted of consulting digital medical records from the electronic system of the public health system (*Sistema Único de Saúde eletrônico* — e-SUS), which is the registration tool used by the RAP. Thus, the follow-up of the cases was assessed and classified by treatment according to the clinical stage of the disease and current care protocol, the Clinical Protocol and Therapeutic Guidelines (*Protocolo Clínico e Diretrizes Terapêuticas* — PCDT) for Comprehensive Care for People with Sexually Transmitted Infections<sup>(5)</sup>. In addition, information concerning the presence of clinical manifestations and follow-up with non-treponemal infection test was added to support the categorization.

The outcome of the AS treatment was categorized as follows:

- Adequate: treatment registered with the appropriate dose of benzathine penicillin within the acceptable time frame (14-day interval tolerance between applications), which matches the clinical stage of AS; alternative treatments (doxycycline and ceftriaxone) associated with a non-treponemal test to observe bacterial load reduction and absence of treatment in case of prior notification of the disease, and non-reactive non-treponemal test or low titrations (≤1:4), according to the recorded clinical history;
- Inadequate: incorrect dose of benzathine penicillin, outside the tolerance range of administration, and with medicines not included in the current care protocols;
- Pending: when there is no record in the information systems about the complete treatment with benzathine penicillin according to the clinical stage of the pathology, or lack of current non-treponemal test results when necessary to confirm the disease, and in the absence of registration of follow-up non-treponemal test, when alternative treatments were performed.

The detection rate in the DA regions was calculated through the number of cases of RT reactive for AS in each DA, divided by the total number of tests performed in the respective administration, multiplied by one hundred. We performed descriptive analyses of all sociodemographic variables investigated with their absolute and relative frequency distributions to build the epidemiological profile.

The process of notification verification was carried out in partnership with the General Coordination of Health Surveillance (*Coordenadoria Geral de Vigilância Sanitária* — CGVS), through the Notifiable Diseases Information System (*Sistema de Informação de Agravos de Notificação* — SINAN), which was consulted to check whether the cases of AS meeting these criteria were included, that is, if the notification was made by the services according to the information notice no. 02-SEI/ 2017<sup>(9)</sup>.

We adopted the chi-square test and linear trend p-value in ordinal categorical variables in a bivariate analysis to verify the association of sociodemographic variables with the treatment outcome, with a 95% significance level, using the statistical software SPSS (IBM SPSS Statistics 22). This analysis combined the treatments considered "Inadequate" and "Pending" to enable the investigation with two outcomes. We conducted a descriptive analysis with the Epi Info software (Epi Info<sup>TM</sup>, CDC). Age groups were classified according to SINAN.

### RESULTS

A total of 16,588 RTs for syphilis were performed in RAP and SAE in the city of Porto Alegre in the first semester of 2018, and 1,817 (11%) of them were reactive for syphilis. The analysis of the study population excluded 353 pregnant women, 7 who became pregnant during the treatment period, and 4 who refused any treatment, resulting in a total of 1,453 patients monitored for AS in the studied period.

According to **Table 1**, the distribution of the 1,453 cases of AS monitored during the study period was not homogeneous in the DA

territories, and the detection rate of syphilis by RT was higher in the Center region.

**Table 2** presents the profile of the patients, showing a slight predominance of males (50.7%), and that 53.6% of the sample declared being white. Most participants belonged to the age group 20 to 29 years (34.1%), with a minimum age of 14 years and maximum of 86 years, and had incomplete elementary school (34.3%). The vast majority of the sample was not homeless (94.9%).

As demonstrated in **Table 3**, the East/Northeast DAs had the highest percentage of adequate treatment. Inadequate treatment was predominant in the Glória/Cruzeiro/Cristal DAs, and pending cases were more frequent in the Center DA, reaching 81.4%. The overall analysis of the city revealed that Porto Alegre has an adequate treatment percentage of 37.6%.

**Table 4** shows an association in the bivariate analysis between treatment outcomes and the sociodemographic variables schooling, gender, age group, and homelessness, and a linear trend in the schooling and age group variables, that is, the higher the schooling and age, the higher the prevalence of adequate treatment. We found no association with the variable ethnicity.

**Table 5** indicates that, although most AS cases are notified to SINAN (52.1%), the number and percentage of cases with no notification is high, even in the National List of Compulsory Notification, despite this communication being mandatory to health professionals.

The detection rate of RT for syphilis was 11% in Porto Alegre during the study period, higher than expected according to the incidence data presented by the Epidemiological Bulletins for the city, and higher than reported by the surveillance regarding syphilis in pregnant women, estimated in  $4\%^{(1,2)}$ .

The analysis per health region showed that the Center DA had a detection rate higher than the one in the city, reaching 14.5%. The administration and policies for prevention, care, and monitoring of syphilis in the city and its territory should take this fact into account. Concomitantly, this DA presented a large percentage (81.4%) of cases with pending treatment, which might be due to the testing of people from other cities who have access to the central region of Porto Alegre, or because this location has the highest concentration of homeless people<sup>(10)</sup>.

The sociodemographic profile found in this study was similar to the analysis based on the compulsory notifications of AS from the Epidemiological Bulletin of Syphilis<sup>(1)</sup>. Most of them have incomplete elementary school, are white, male, and aged between 20 and 29 years. However, with respect to gender, this bulletin reveals a

**Table 1** – Number of reactive cases for acquired syphilis (n=1,453), tests performed (n=16,588), and detection rate according to District Health Administrations in Porto Alegre in the first semester of 2018.

District Administration	Reactive n	Tests n	Detection Rate
Center	285	1,908	14.9
Glória/Cruzeiro/Cristal	251	2,874	8.7
East/Northeast	155	1,919	8.1
North/Eixo Baltazar	126	1,998	6.3
Northwest/Humaitá/ Navegantes/Islands	112	1,694	6.6
Partenon/Lomba do Pinheiro	292	3,200	9.1
Restinga/Far South	113	1,365	8.3
South/South Center	119	1,630	7.3

difference of 1.5 cases in males for each case in females, while in the present study, the proportion was practically equal between genders; nonetheless, the analysis showed statistical significance, with prevalence of females in cases with adequate treatment.

The minimum age of 14 years in reactive tests for syphilis is worrying and reinforces the need for preventive actions against STIs among adolescents. The group in question belongs to a priority segment targeted by the Syphilis No project. This project is a partnership between the Ministry of Health and the Universidade Federal do Rio Grande do Norte that aims to expand the population's access to diagnosis and treatment in the basic health network. The initiative presents a wide range of strategies to disseminate information, achieve its purposes, and reach the priority populations (young people, pregnant women, sexually active population, and health professionals)<sup>(11)</sup>.

Homeless people represented 5.1% of the cases of RT reactive for syphilis, showing the importance of care focused on social vulnerability. The study corroborates this issue due to the association of adequate treatment and not being homeless. Porto Alegre has 2,115 homeless adults, according to research conducted by the Universidade Federal do Rio Grande do Sul (UFRGS), representing 0.1% of the population of the city<sup>(10)</sup>.

**Table 2** – Sociodemographic variables of individuals monitored for acquired syphilis (n=1,453) in the primary care network and specialized care services of Porto Alegre in the first semester of 2018.

Variable	N (%)
Schooling	
Illiterate	23 (1.6)
Incomplete elementary school	499 (34.3)
Complete elementary school	224 (15.4)
Incomplete high school	250 (17.2)
Complete high school	309 (21.3)
Incomplete higher education	70 (4.8)
Complete higher education	78 (5.4)
Ethnicity	
White	779 (53.6)
Black	404 (27.8)
Multiracial	239 (16.4)
Indigenous	12 (0.8)
Asian	8 (0.6)
Unknown	11 (0.8)
Gender	
Female	716 (49.3)
Male	737 (50.7)
Age group (years)	
10 to 14	7 (0.5)
15 to 19	173 (11.9)
20 to 29	496 (34.1)
30 to 39	316 (21.8)
40 to 49	197 (13.6)
50 to 59	150 (10.3)
60 to 69	89 (6.1)
70 or older	25 (1.7)
Mean (±standard deviation) (minimum; maximum)	34.6(±14.4) (14.86)
Homeless	
No	1,379 (94.9)
Yes	74 (5.1)

The bivariate analysis also evidences the association of vulnerability with the treatment outcome through the schooling variable, showing a linear relationship between better education and the greater frequency of adequate treatment. In Brazil, schooling is often used as a proximal income. This fact may indicate another aspect of vulnerability associated with treatment in this study. According to the information collected by the Institute of Applied Economic Research (*Instituto de Pesquisa Econômica Aplicada* — IPEA), the main factor of expansion in working income was the increase in schooling<sup>(12)</sup>. **Table 5** – Distribution of cases meeting the criteria for mandatory notifications (n=432) among individuals monitored for acquired syphilis in the primary care network and specialized care services of Porto Alegre in the first semester of 2018.

Notified to SINAN	n (%)
Yes	225 (52.1)
No	207 (47.9)

SINAN: Notifiable Diseases Information System (Sistema de Informação de Agravos de Notificação).

Table 3 – Treatment outcome of monitored cases of acquired syphilis according to District Administrations in Porto Alegre in the first semester of 2018.

	Adequate n (%)	Inadequate n (%)	Pending (%)	Total n (100%)
District Administration				
Center	52 (18.2)	1 (0.4)	232 (81.4)	285
Glória/Cruzeiro/Cristal	112 (44.6)	7 (2.8)	132 (52.6)	251
East/Northeast	72 (46.5)	3 (1.9)	80 (51.6)	155
North/Eixo Baltazar	56 (44.4)	0 (0)	70 (55.6)	126
Northwest/Humaitá/Navegantes/Islands	41 (36.6)	1 (0.9)	70 (62.5)	112
Partenon/Lomba do Pinheiro	115 (39.4)	1 (0.3)	176 (60.3)	292
Restinga/Far South	47 (41.6)	4 (3.5)	62 (54.9)	113
South/South Center	51 (42.9)	5 (4.2)	63 (52.9)	119
Total of Porto Alegre	546 (37.6)	22 (1.5)	885 (60.9)	1,453

**Table 4** – Prevalence of adequate (n=546) and inadequate/pending (n=907) treatments for acquired syphilis according to sociodemographic variables of monitored cases in Porto Alegre in the first semester of 2018.

Variable	Adequate n (%)	Inadequate/Pending n (%)	P*
Schooling			
Illiterate	9 (1.6)	14 (1.5)	
Incomplete elementary school	201 (36.8)	298 (32.9)	
Complete elementary school	84 (15.4)	140 (15.5)	
Incomplete high school	99 (18.1)	151 (16.6)	0.001
Complete high school	123 (22.5)	186 (20.5)	
Incomplete higher education	15 (2.8)	55 (6.1)	
Complete higher education	15 (2.8)	63 (6.9)	
Ethnicity			
White	265 (48.5)	514 (56.6)	
Black	170 (31.2)	234 (25.8)	
Multiracial	99 (18.1)	140 (15.4)	0.000
Indigenous	5 (0.9)	7 (0.8)	0.098
Asian	3 (0.6)	5 (0.6)	
Unknown	4 (0.7)	7 (0.8)	
Gender			
Female	295 (54)	421 (46.4)	0.005
Male	251 (46)	486 (53.6)	0.005
Age group (years)			
10 to 14	3 (0.5)	4 (0.4)	
15 to 19	79 (14.5)	94 (10.4)	
20 to 29	185 (33.9)	311 (34.3)	
30 to 39	85 (15.6)	231 (25.5)	-0.001
40 to 49	70 (12.8)	127 (14)	<0.001
50 to 59	59 (10.8)	91 (10)	
60 to 69	51 (9.3)	38 (4.2)	
70 or older	14 (2.6)	11 (1.2)	
Homeless			
No	533 (97.6)	846 (93.3)	-0.001
Yes	13 (2.4)	61 (6.7)	<0.001

\*P for heterogeneity of proportions (dichotomous variable) and linear trend (ordinal variable).

The 2010 IBGE census<sup>(13)</sup> revealed that Porto Alegre has a black population of 10.2%, multiracial of 10%, and indigenous of 0.23%. These same population segments presented respectively 27.8, 16.4, and 0.8% of RT reactive for AS. This information shows an increased risk of a reactive result for this pathology, again reinforcing the need for attention to social vulnerability in the follow-up of syphilis cases. However, this study found no association between the variable ethnicity and the adequate treatment outcome.

We emphasize that, although the incidence of syphilis is high in Porto Alegre, as published in epidemiological bulletins, cases are still underreported, and in the studied period — one semester — , 47.9% of cases had not been notified despite meeting the notification criteria for AS.

### CONCLUSION

This study identified a higher than expected RT detection for syphilis, which corroborates data from epidemiological bulletins showing an elevated incidence of AS in Porto Alegre. In addition, some DAs of the city had higher detection rates, despite the presence or absence of adequate treatment in the regions. This information enables a situation diagnosis of the city and can support management actions and public policies, while Brazil focuses on the Syphilis No program.

The development of actions to structure and organize the processes for AS monitoring becomes indispensable, especially due to the evidence of an association between sociodemographic variables that indicate social vulnerability and the adequate treatment outcome. We underline that the diagnosis and treatment of people with STIs and their sexual partners break the transmission chain, preventing possible complications. This is the only way we can fight a disease that has no vaccine but can be treated and cured.

Based on this study, we propose to continue monitoring the cases to expand and improve information about syphilis aggravation and management. We also suggest adding other variables related to income, sexual orientation, and gender identity to this analysis information, which would enable other preventive strategies.

### Participation of each author

Gustavo Haas Lermen – Main author. Lisiane Morelia Weide Acosta – Advisor. Fabiana Ferreira dos Santos – Co-advisor. Juanita Natasha Garcia de Oliveira – Co-author.

### Funding

This study was funded by the authors.

### **Conflict of interests**

The authors declare no conflict of interests.

### **Approval by the Human Research Ethics Committee**

The Research Ethics Committee of the Municipal Health Department of Porto Alegre approved this study under protocol no. 2931412.

### REFERENCES

- Brasil. Boletim Epidemiológico Sífilis 2017 [Internet]. Brasil: Ministério da Saúde; 2017 [cited on Apr. 23, 2018]. Available at: http://www.aids. gov.br/pt-br/pub/2017/boletim-epidemiologico-de-sifilis-2017
- Porto Alegre. Boletim Epidemiológico nº 67 [Internet]. Porto Alegre: Coordenadoria Geral de Vigilância em Saúde; 2017 [cited on Dec. 15, 2018]. Available at: http://lproweb.procempa.com.br/pmpa/prefpoa/cgvs/ usu\_doc/boletim67\_sifilis.pdf
- Porto Alegre. Boletim Epidemiológico nº 69 [Internet]. Porto Alegre: Coordenadoria Geral de Vigilância em Saúde; 2018 [cited on Dec. 18, 2018]. Available at: http://lproweb.procempa.com.br/pmpa/prefpoa/cgvs/ usu\_doc/boletimepidemiologico-cgvs-sms-pmpa-69.pdf
- Campos ALA, Araújo MAL, Melo SP, Gonçalves MLC. Epidemiologia da sífilis gestacional em Fortaleza, Ceará, Brasil: um agravo sem controle. Cad Saúde Pública. 2010;26(9):1747-55. http://dx.doi.org/10.1590/ S0102-311X2010000900008
- Brasil. Protocolo Clínico e Diretrizes Terapêuticas (PCDT) para Atenção Integral às Pessoas com Infecções Sexualmente Transmissíveis (IST) [Internet]. Brasil: Ministério da Saúde; 2016 [cited on Dec. 20, 2018]. Available at: http://portaldeboaspraticas.iff.fiocruz.br/wp-content/ uploads/2019/02/pcdt\_ist\_para\_web\_-\_nao\_diagramado1.pdf
- Porto Alegre. Nota Técnica SMS nº 001/2018. Diagnóstico e tratamento de sífilis na rede de saúde de Porto Alegre [Internet]. Porto Alegre: Secretaria Municipal de Saúde; 2018 [cited on Jan. 13, 2019]. Available at: http://dopaonlineupload.procempa.com.br/dopaonlineupload/2511\_ ce\_226208\_3.pdf
- Porto Alegre. Protocolo de Assistência ao Pré-Natal de Baixo Risco [Internet]. Porto Alegre: Secretaria Municipal de Saúde; 2015 [cited on Jan. 7, 2019]. Available at: http://lproweb.procempa.com.br/pmpa/ prefpoa/sms/usu\_doc/protocolo\_pre\_natal\_2015.pdf
- Porto Alegre. Estrutura [Internet]. Porto Alegre: Secretaria Municipal de Saúde; 2017 [cited on Jan. 8, 2018]. Available at: http://www2.portoalegre. rs.gov.br/sms/default.php?p\_secao=808&gt
- Brasil. Nota informativa nº 2-SEI/2017. Altera os Critérios de Definição de Casos para Notificação de Sífilis Adquirida, Sífilis em Gestantes e Sífilis Congênita [Internet]. Brasil: Ministério da Saúde; 2017 [cited on May 2, 2018]. Available at: http://www.aids.gov.br/pt-br/legislacao/notainformativa-no-02-sei2017-diahvsvsms
- Universidade Federal do Rio Grande do Sul. Cadastro e mundo da população adulta em situação de rua de Porto Alegre/RS [Internet]. Porto Alegre: Universidade Federal do Rio Grande do Sul; 2016 [cited on Jan. 31, 2018]. Available at: http://www2.portoalegre.rs.gov.br/fasc/default. php?reg=41&p\_secao=120
- Brasil. [Internet]. Lembre de se cuidar: sífilis teste, trate e cure. Brasil; Ministério da Saúde; 2018 [cited on Jan. 1, 2019]. Available at: http:// www.sifilisnao.com.br/index\_desktop.html#lembretejovens
- Brasil. Aumento da escolaridade amplia renda do trabalho [Internet]. Instituto Brasileiro de Pesquisa Econômica Aplicada; 2013 [cited on Feb. 4, 2019]. Available at: http://www.ipea.gov.br/portal/index. php?option=com\_content&view=article&id=20067
- Brasil. Instituto Brasileiro de Geografia e Estatística. Portal. [Internet]. Instituto Brasileiro de Geografia e Estatística; 2010 [cited on Jan. 15, 2018]. Available at: https://www.ibge.gov.br/

### Address for correspondence: GUSTAVO HAAS LERMEN

Rua Getúlio Vargas, 1.035/407 – Bairro Menino Jesus Porto Alegre (RS), Brazil CEP: 90150-001 E-mail: gustavolermen@mx2.unisc.br

Received on: 07.15.2019 Approved on: 09.03.2019

# PROGRAM FOR PREVENTION OF Mother-to-Child Transmission of Syphilis and HIV in Brazil: missed opportunities

### Programa de prevenção da transmissão materno-infantil de sífilis e HIV no Brasil: oportunidades perdidas

Angelica Espinosa Miranda<sup>1</sup> , Mariangela Freitas da Silveira<sup>2</sup>, Maria Alix Leite Araujo<sup>3</sup>, Leonor de Lannoy Tavares<sup>4</sup>, Sandra Fagundes Moreira-Silva<sup>5</sup>, Leila Cristina Silva<sup>6</sup>, Adele Schwartz Benzaken<sup>7</sup>, Valeria Saraceni<sup>8</sup>

### ABSTRACT

While antenatal screening for HIV and syphilis is part of the national policy in Brazil, screening and treatment coverage remain inadequate in many parts of the country. This study aimed to describe missed opportunities concerning mother-to-child transmission (MTCT) from the point-of-view of pregnant women, health professionals, and health care managers. A semi-structured interview was conducted in six Brazilian states. Pregnant women, health professionals, and health care managers were interviewed to identify failures in the process of care for pregnant women and MTCT of syphilis or HIV. The project had a quantitative approach but included open-ended questions to capture the views of participants regarding the feasibility of strategies adopted for controlling MTCT. The sample consisted of 109 women, 62 health professionals, and 34 health care managers. The median age of women was 24 (range: 15–46) years, and the median schooling was 8 years. Eighty percent of those interviewed received prenatal care. Among those who attended antenatal visits, the median was 6.43 (range: 1–20) visits. Managers and health professionals had a median of 10 (range: 4–25) working years. In the interviews, the managers declared that they had provided tests and treatment for these infections, but health professionals stated that they did not have tests or treatment available to offer, and most women complained about the difficulties of receiving treatment. Organizing the logistics and breaking down barriers related to care in Brazil is challenging. An adequate health care system and policy factors that address this situation can help to eliminate MTCT by implementing strategies adopted to control these infections in the country.

Keywords: Mother-to-child transmission; HIV; syphilis; health system; health policy.

### RESUMO

Embora o rastreamento para HIV e sífilis no pré-natal faça parte da política nacional no Brasil, a cobertura do rastreamento e tratamento permanece inadequada em muitas partes do país. O objetivo deste estudo foi descrever as oportunidades perdidas de transmissão materno-infantil (TMI) do ponto de vista de gestantes, profissionais de saúde e gestores de saúde. Uma entrevista semiestruturada foi realizada em seis estados brasileiros. Foram entrevistadas gestantes, profissionais de saúde e gestores dos serviços de saúde, com o objetivo de identificar falhas no processo de atendimento às gestantes e à TMI de sífilis ou HIV. A abordagem do projeto foi quantitativa, mas perguntas abertas foram incluídas para capturar as opiniões dos participantes sobre a viabilidade das estratégias adotadas para o controle da TMI. Participaram do estudo 109 mulheres, 62 profissionais de saúde e 34 gestores. A mediana de idade das mulheres foi de 24 (intervalo:15-46) anos e a mediana de escolaridade foi de 8 anos. Oitenta por cento dos entrevistados fizeram consultas de pré-natal. Entre as que participam de consultas pré-natais, a mediana foi de 6,43 (intervalo: 1 a 20). Gestores e profissionais de saúde tiveram uma mediana de 10 anos de trabalho (intervalo: 4-25). Nas entrevistas, os gestores disseram que haviam fornecido testes e tratamento para essas infecções, mas os profissionais de saúde disseram que nem sempre tinham testes ou tratamentos disponíveis para oferecer às pacientes e a maioria das parturientes reclamou das dificuldades em receber tratamento. Organizar a logística e derrubar barreiras de cuidado ainda representam um desafio no Brasil. O sistema de saúde com funcionamento adequado e uma ação política de enfrentamento da situação podem ajudar a eliminar a TMI, quando atuam na aplicação das estratégias adotadas pelo país no controle dessas infecções.

Palavras-chave: Transmissão da mãe para o filho; HIV; Sífilis; Sistema de saúde; Política de saúde

### INTRODUCTION

Brazil is a multifaceted country experiencing complex economic, social, and environmental changes. Social, economic, and regional inequalities are still extreme and rampant; basic living conditions must be dramatically improved for a large part of the population. Health problems are often a result of social and environmental changes and remain unabated<sup>(1)</sup>.

Sexually transmitted infections (STIs), including HIV infection and syphilis, are important public health issues that lead to perinatal morbidity and mortality<sup>(2,3)</sup>. Neonates who survive with HIV or congenital syphilis are at a greater risk of low birth weight, premature delivery, congenital anomalies, and long-term sequelae — such as deafness

<sup>&</sup>lt;sup>1</sup>Graduate Program in Collective Health, Universidade Federal do Espírito Santo – Vitória (ES), Brazil.

<sup>&</sup>lt;sup>2</sup>Graduate Program in Epidemiology, Universidade Federal de Pelotas – Pelotas (RS), Brazil.

<sup>&</sup>lt;sup>3</sup>Graduate Program in Collective Health, Universidade de Fortaleza – Fortaleza (CE), Brazil.

<sup>&</sup>lt;sup>4</sup>Secretariat of Health of Distrito Federal – Brasília (DF), Brazil.

<sup>&</sup>lt;sup>5</sup>STD/AIDS State Program, State Secretariat of Health of Espírito Santo – Vitória (ES), Brazil.

<sup>&</sup>lt;sup>6</sup>Health Surveillance Foundation, State Secretariat of Health of Amazonas – Manaus (AM), Brazil.

<sup>&</sup>lt;sup>7</sup>Aids Health Care Foundation – Manaus (AM), Brazil.

<sup>&</sup>lt;sup>8</sup>Health Surveillance Administration, Municipal Secretariat of Health, and Civil Defense of the State of Rio de Janeiro - Rio de Janeiro (RJ), Brazil.

and neurological impairment, for example<sup>(3,4)</sup>. These consequences are preventable if health clinics are prepared to provide adequate care, and if infected mothers are identified earlier during pregnancy and receive proper treatment<sup>(5,6)</sup>. In this process, antenatal HIV and syphilis testings are essential to prevent the mother-to-child-transmission (MTCT) of these infections and offer an opportunity for women and other family members to find out their status and access treatment<sup>(7,8)</sup>.

Prevention of MTCT of HIV and syphilis are significant and recognized steps towards achieving the United Nations Sustainable Development Goals, particularly the health-related goals of decreasing pregnancy complications and child and maternal mortality<sup>(9)</sup>. In 2010, the Pan American Health Organization (PAHO) countries agreed to implement strategies for eliminating MTCT of syphilis and HIV by 2015. HIV elimination goals consisted of a reduction in the incidence of HIV cases in children to 0.3 cases/1,000 live births and an MTCT rate  $\leq 2\%$ ; syphilis elimination goals included prevalence of 0.5 cases/1,000 live births. In 2016, these goals were changed and extended upon by the Plan of Action for the Prevention and Control of HIV and Sexually Transmitted Infections 2016–2021<sup>(10)</sup>.

While antenatal screening for HIV and syphilis are part of Brazil's national policy, both screening and treatment coverage remain inadequate in many parts of the country, mainly due to technical and logistical problems. Together, the health care system and new policies are capable of eliminating MTCT while overcoming barriers and integrating surveillance, as well as strengthening service delivery and community leadership<sup>(11)</sup>.

In Brazil, an estimated 0.38% of all pregnant women are infected with HIV, corresponding to almost 11,000 pregnant women infected with HIV per year. A total of 108,134 cases of pregnant women with HIV were notified from 2000 to June 2017. Over ten years, the detection of HIV in pregnant women increased 23.8%<sup>(12)</sup>. In 2016, Brazil had 87,593 cases of acquired syphilis, 37,436 cases of syphilis in pregnant women, and 20,474 cases of congenital syphilis, with 185 deaths among them. The detection rate of syphilis in pregnant women and the incidence rate of congenital syphilis were 12.4/1,000 live births, respectively<sup>(13)</sup>.

A previous study in Brazil evaluated the cascade of HIV care in pregnant women and identified flaws in all stages of care. It showed a lack of connection between primary care and referral centers for HIV/AIDS, which could organize the care delivery for families and promote better outcomes for children<sup>(14)</sup>. Another study described increased detection rates of syphilis in pregnancy and revealed that better organization of health services and professional awareness could improve the care offered to these patients<sup>(15)</sup>.

In Brazil, both prenatal care and childbirth coverage are close to 100%; syphilis and HIV tests and drugs are available for free to prevent MTCT. Formula for HIV-exposed children is also available. It is crucial to identify the gaps that could help eliminate these infections in Brazil.

### **OBJECTIVE**

This study aimed to describe the missed opportunities in preventing MTCT in Brazil based on the perspective of women in the maternal ward who delivered babies with HIV and/or syphilis, health professionals, and health care managers working in the same institutions.

### METHODS

In 2015, a sample consisting of cases of pregnant women notified for syphilis and/or HIV with a positive newborn was chosen in the state capital and one city in six Brazilian states: Amazonas, Ceará, Espírito Santo, Rio de Janeiro, Rio Grande do Sul, and the Federal District. Participants were equally chosen among states. The selection was based on the availability of complete data in the Notifiable Diseases Information System (Sistema de Informação de Agravos de Notificação — SINAN) form from the mother and her child. The maternity hospitals involved presented different characteristics, according to the states where they belonged, but all of them were part of Brazil's public health care system (Sistema Único de Saúde - SUS). MTCT cases were chosen from the SINAN database, and women were invited to participate in the interview. Physicians, nurses, and managers working in the same institution where these cases happened were also invited to participate in the interview and share their views on MTCT of HIV and syphilis in their health care center and city.

A semi-structured interview, including demographical (age, gender, education, occupation, time working in the field, etc.) and health care access data (prenatal care, access to HIV/syphilis diagnosis and treatment, follow-up, professional years of experience, training opportunities for professionals, etc.), was conducted with 20 women aged 15 years or over, who delivered their babies at the chosen health center, 10 health professionals, and five health care center managers in each Brazilian state. The data collected were entered in an Excel worksheet (Microsoft<sup>®</sup>) and analyzed in SPSS 19.0. We used measures of central tendency and dispersion for continuous variables and measures of frequency for categorical variables to build the participants' profile.

A smaller number of women in the maternal ward who delivered babies with HIV and/or syphilis, health professionals, and health care managers working in the same institutions were invited for an in-depth interview focusing on why they thought the transmission had occurred and what were the failures. We opted for the individual interview because it is a faster method for obtaining qualitative data, and also more suitable for this project, since participants had little free time to answer the questions. Moreover, interviews allow people to talk on their own terms. Interviews took place at private venues chosen by participants and lasted 30-40 minutes; they were recorded and later transcribed and de-identified for analysis. We included open-ended questions to understand better how HIV and syphilis counseling and diagnosis were carried out at health care centers and grasp the opinions of women, health professionals, and managers about health care access and approach, including laboratory and medical supplies and professional training. These answers would offer better knowledge and the possibility of reassessing the relevance of current strategies or the control of MTCT of HIV and syphilis.

From the in-depth interviews, we extracted and included quotes to complement the descriptive approach and provide a better understanding of the missed opportunities to prevent MTCT of HIV and syphilis. Interviews focused on the participants' perceptions of existing workflows between diagnosis and care delivered to each case, and how they reacted to them. With this technique, we were able to find more about the words they use, their priorities and concerns<sup>(16)</sup> regarding MTCT, and its implications for health care units. It could offer answers on how to improve services and professional approach.

Interviews were conducted after the participants invited signed an informed consent form. We assigned names to the participants to preserve their anonymity and confidentiality. The Research Ethics Committee from the Universidade Federal do Espírito Santo approved the project (#640,580/2014). Privacy and confidentiality were protected in all phases of the project using data codification.

### RESULTS

A total of 109 parturient women, 62 health professionals, and 34 health care managers were interviewed. The median age of parturient women was 24 years, ranging from 15 to 46, and the median schooling was 8 years. Among those who attended antenatal visits, the median was 6.4 (range: 1–20) visits. Managers and health professionals had a median of 10 (range: 4–25) working years in the field.

 Table 1 describes the perception of care by parturient women infected with HIV and syphilis. We highlight that 62.4% of them

**Table 1** – Perception of the health care system reported by women who delivered babies with HIV and/or syphilis in Brazil, 2014.

,	,
Variables	N (%)
Considered health professionals prepared to counsel the	em about HIV
and syphilis during antenatal care	
No	26 (23.8)
Yes	68 (62.4)
Did not attend antenatal care	15 (13.8)
Quality care in the maternity hospital	
Excellent/very good	31 (28.4)
Good	63 (57.9)
Bad	15 (13.7)
Received counseling about diagnosis in the hospital	
No	23 (21.1)
Yes	86 (78.9)
Who counseled you about the infections in the hospital	?
Health team (physician and nurse)	28 (25.7)
Physician alone	42 (38.5)
Nurse alone	31 (28.4)
No one	8 (7.3)
Did you receive treatment for the infection?	
No	9 (8.2)
Yes	100 (91.8)
Did the baby receive care for PMTCT?	
No	7 (6.4)
Yes	102 (93.6)
Did health professionals advise you to talk to your partr	ner?
No	14 (12.8)
Yes	95 (87.2)
How did you talk to your partner about the diagnosis?	
He already knew it	25 (22.9)
We were together in the center when the health professional disclosed the diagnosis	51 (50.8)
I decided not to talk about it with my partner	23 (21.1)
I do not have any contact with my partner anymore	10 (9.2)

HIV: human immunodeficiency virus; PMTCT: prevention of mother-tochild transmission. considered health professionals prepared to counsel them about HIV and syphilis during antenatal care. A total of 21.1% decided against talking about their diagnosed infections with their partners.

**Table 2** presents the characteristics and perspectives of health professionals involved in HIV and syphilis cases. Interestingly,

**Table 2 –** Profile and perspectives of health professionals involved in the care of cases of mother-to-child transmission of HIV and syphilis in Brazil, 2014.

Variables	N (%)
Professional age	
<30	11 (17.7)
30–35	8 (12.9)
>35	43 (69.4)
Profession	
Physician	28 (45.2)
Nurse	31 (50.0)
Social worker	2 (3.2)
Psychologist	1 (1.6)
Years in professional practice	
<5	12 (19.4)
5–9	12 (19.3)
10–15	11 (17.7)
>15	27 (43.5)
Received training for PMTCT of HIV and syphilis	
No	32 (51.6)
Yes	30 (48.4)
Would like to be trained for PMTCT of HIV and syphilis	
No	13 (21.0)
Yes	49 (79.0)
Syphilis diagnosis available in your health center	- ( )
Point-of-care test	25 (40.3)
VDRL	34 (54.8)
Both tests	3 (4.8)
HIV diagnosis available in your health center	- ( - )
Point-of-care test	57 (91.9)
Elisa	5 (8.1)
How long it takes to receive the HIV test results?	- (- )
Less than 1 hour	40 (64.5)
1–24 hours	17 (27.4)
2–7 davs	1 (1.6)
8 or more days	4 (6.4)
How long it takes to receive the syphilis test results?	
Less than 1 hour	19 (30.6)
1–24 hours	27 (43.5)
2–7 davs	7 (11.3)
8 or more days	9 (14.6)
Have access to confirmatory tests for HIV and syphilis	- (1110)
No	11 (17.7)
Yes	51 (82.3)
Have regular access to diagnostic tests for HIV and syph	ilis
No	5 (8.1)
Yes for both infections	52 (83 9)
Yes for HIV but not for syphilis	5 (8.0)
Medicines available to prevent MTCT of HIV and synhilis	0 (0.0)
Yes for both infections	57 (91 9)
Yes for HIV but not always for synhilis	5 (8 1)

HIV: human immunodeficiency virus; MTCT: mother-to-child transmission; PMTCT: prevention of MTCT; VDRL: venereal disease research laboratory; Elisa: enzyme-linked immunosorbent assay.

despite almost 50% of them having declared being trained in MTCT of HIV and syphilis, 79% reported that they would like to receive training in the prevention of MTCT. **Table 3** shows the characteristics and perspectives of health care managers involved in the HIV and syphilis cases included in this study; 61.8% reported that they had been in their current management position for less than five years, and less than 50% had received any kind of management training.

In the in-depth interviews, we tried to understand the offer of diagnosis and treatment in the units as a way to measure how the managers and health professionals perceive the availability of supplies, training, clinical support, and supervision. The managers declared that they had provided tests and treatment for these infections in their health centers. Some answers are quoted below:

I cannot explain this! We buy tests and antibiotics. We also provide training for health professionals. What more we can do? (Ana, 36)

I know there are problems! It is a complicated situation. We try to do our best, but sometimes things do not work out the way they should! (Jorge, 42)

However, health professionals stated that they did not always have tests or treatment available to offer to their patients. They also complained that they did not feel confident enough to perform rapid tests and that they would like to receive more training. Training and supplies sometimes were missing.

> Sometimes there are no rapid tests at the center, and I need to recommend regular tests, but Elisa (HIV) or VDRL (Syphilis) results can take at least a week. (Miguel, 28)

> I would like to receive more training to feel more confident in carrying out rapid tests. I am still doubtful about how to use it. They are treponemal tests, how can I know if it is not just a scar? (Elisa, 33)

> When the pregnant woman tests positive, I offer counseling and send her on to treatment. I cannot administer penicillin in my center because conditions are not safe, and pregnant women must seek out another health center. (Paulo, 45)

In the interviews with the women, the main goal was to discuss access and barriers to receive appropriate and comprehensive care. Most parturient women complained about how hard it was to understand the disease and receive treatment. It is hard to have effective care if you do not understand the problem you have to handle.

> The doctor was nice to me. He told me I needed to receive treatment for syphilis to try to protect my baby, but he did not explain anything about the disease to me. What is syphilis? I was wondering what it means. (Maria, 22)

> The health center did not have penicillin, and they sent me to the hospital, but I did not go because I did not have

**Table 3 –** Profile and perspectives of health care managers in centers that treated cases of mother-to-child transmission of HIV and syphilis in Brazil, 2014.

and opping in Diana, 2011	
Variables	N (%)
Professional age	
<30	3 (8.8)
30–35	8 (23.5)
>35	23 (67.6)
Professional background	
Physician	13 (38.2)
Nurse	16 (47.1)
Other professions	5 (14.6)
Years working as a manager	
<5	21 (61.8)
5–9	6 (17.6)
10–15	4 (11.8)
>15	3 (8.8)
Received management training for health care service	S
No	19 (55.8)
Yes	15 (44.2)
Received management training for PMTCT of HIV and	l syphilis
No	17 (50.0)
Yes	17 (50.0)
Thinks that management training for PMTCT of HIV ar	nd syphilis is
necessary	
No	15 (44.1)
Yes	19 (55.9)
HIV and syphilis tests available in the health centers	
Point-of-care for both infections	12 (35.3)
VDRL for syphilis and Elisa for HIV	5 (14.7)
Point-of-care for HIV and VDRL for syphilis	17 (50.0)
How long it takes to receive the HIV test results?	
Less than 1 hour	23 (67.6)
1–24 hours	4 (11.8)
2-7 days	3 (8.8)
8 or more days	4 (11.8)
How long it takes to receive the syphilis test results?	
Less than 1 hour	14 (41.2)
1–24 hours	10 (29.4)
2-7 days	4 (11.8)
8 or more days	6 (17.6)
Have access to confirmatory tests for HIV and syphilis	0 (00 5)
NO No -	8 (23.5)
Yes	26 (76.5)
Have regular access to diagnostic tests for HIV and sy	
Yes for both infections	30 (88.2)
Yes for HIV but not for syphilis	4 (11.8)
Medicines available to prevent MICI of HIV and syph	
Yes for both infections	31 (91.2)
Yes for HIV but not always for syphilis	3 (8.8)
Peniciliin available in antenatal clinic and nospital	
No	4 (11.8)
Yes	30 (88.2)
Antiretroviral drugs available in antenatal clinic and ho	spital
NO	2 (5.9)
	32 (94.1)
i ne unit has a program to follow patients with positive	tests
NO	21 (61.8)
Yes	13 (38.2)

HIV: human immunodeficiency virus; MTCT: mother-to-child transmission; PMTCT: prevention of MTCT; VDRL: venereal disease research laboratory; Elisa: enzyme-linked immunosorbent assay.

money to pay for the bus fare, and I could not lose another day of work. (Helena, 26)

I did not tell my partner about the diagnosis because I was afraid he would beat me. I thought it was enough to do the treatment alone. (Marta, 31)

It was important to listen to the different health care actors and observe the importance of a better linkage among managers, health professionals, and the client's needs to offer an effective health care access for pregnant women. As important as access to a health unit is to receive the right diagnosis, treatment, and counseling.

### DISCUSSION

The findings suggest problems in the communication of all parties involved in antenatal care and maternal wards, with different information obtained depending on who is answering the questions. They may also indicate inadequate approaches to training and supervision, as per the answers from both managers and health professionals. Although prenatal care is available for almost all pregnant women in Brazil and the public health system offers universal screening for HIV and syphilis, our data showed that broken communication between health care providers and the women admitted for care leads to technical and logistical challenges, as expressed during the in-depth interviews. Women's needs may be better fulfilled with more comprehensive care. The Brazilian Ministry of Health distributes more than 12 million HIV rapid tests and more than 9 million syphilis rapid tests per year to the 27 states<sup>(12,13)</sup>. The states are responsible for the logistics to reach all their cities. Cities, in turn, have health centers to deliver care to the population, including prenatal care. Knowledge and communication in the health system are important tools for organizing services and adequate care<sup>(17)</sup>.

In 2014, the World Health Organization (WHO) presented a guide that included joined processes and criteria for validation of EMTCT (elimination of MTCT) of HIV and syphilis(18); one year later, Cuba became the first country to achieve EMTCT of both HIV and syphilis<sup>(19)</sup>. Other countries have followed Cuba and, in 2016, Thailand and Belarus were validated for EMTCT of HIV and syphilis, Moldova for EMTCT of syphilis, and Armenia for EMTCT of HIV<sup>(20)</sup>. Several countries have EMTCT plans in place, including Brazil, and others are set for EMTCT validation by a target date<sup>(6,18)</sup>. The countries that have achieved EMTCT have similar strategies established to ensure universal and equitable antenatal care services, which include HIV and syphilis testing and treatment at no cost to pregnant women<sup>(21)</sup>. Brazil has the same strategies of these countries, but being a vast country with a large population and huge regional inequalities has hampered the efforts to reach EMTCT. Policy makers and stakeholders must develop strategies according to the characteristics of each region. All regions have shown a combination of strategies referring to social, economic, and regional policies to promote development in a more widespread way over the last few years; however, limitations to policies implemented for the universalization of the health system remain<sup>(22)</sup>.

The study was carried out during a period when the penicillin was not available in most Brazilian health centers, due to the global shortage of the drug. Nevertheless, even when the drug is available, resistance to the intramuscular administration of penicillin at primary care level is still present in health centers across Brazil and should be addressed as a priority to eliminate MTCT of syphilis. The literature has good evidence that penicillin is safe to use in health facilities<sup>(23)</sup>. Regarding HIV infection, no one complained about drug availability or problems with antiretroviral adherence. Prevention of MTCT has been an important step in inhibiting HIV infection in children. However, improving the uptake of and adherence to care continues to be a significant challenge in resource-poor settings<sup>(24)</sup>.

We also found that, at times, pregnant women did not receive appropriate counseling. A significant factor that prevents Brazil from achieving EMTCT of HIV and syphilis is the quality of prenatal care. Despite efforts to improve obstetric and neonatal quality of care, this topic remains a challenge in the Brazilian public health system, both in terms of enhancing its quality as such, and of introducing changes to the principles of care<sup>(25)</sup>. Studies that have evaluated prenatal work processes demonstrated that not all recommended practices are followed, and some are far below the desired, including the time of prenatal care initiation<sup>(26-28)</sup>. Moreover, there are social and economic problems influencing the provision of care. Although health care is considered universal in Brazil through SUS, some women lack funds to reach the health centers, do not have help to leave their older children while away, or cannot miss work<sup>(1,26)</sup>. We highlight that providing free and easy access to health units for prenatal care is not enough if the clinics do not offer adequate diagnosis and treatment. Public health in Brazil needs to improve the dialog among the different care actors. Managers, health professionals, and pregnant women need to be included in the decisions and solutions because it is the only way to have effective control of MTCT cases in Brazil.

In addition, the inadequacy of national registries has also been pointed out as one of the obstacles to improving the quality of prenatal care, as missing and partial information will mislead policy makers and managers in health planning. The government needs to know the number of cases to be able to buy the necessary test and drug supplies. There is also a long way to go to incorporate evaluation as a systematic activity, allowing managers to intervene and redirect actions(20). Therefore, improving training in both care and surveillance, as well as monitoring health professionals in charge of prenatal care through medical record review is paramount to reach EMTCT goals. Domingues et al.29 assessed knowledge, practices, and attitudes of health professionals delivering prenatal care in the city of Rio de Janeiro, and found that lack of knowledge about protocols and limitations regarding management of patients with STI were among the main barriers to dealing with the burden of syphilis. The present study showed that less than 50% of health professionals considered themselves well trained in MTCT of HIV and syphilis, and almost 80% would like to be trained again.

One of the limitations of our study was the small sample size for the descriptive analyses. However, our intention was not to reach statistical power but rather to understand who the professionals in the states are, and what their perceptions regarding the MTCT program are. Face-to-face and audio-recorded interviews are always complicated, and we cannot rule out the possibility of response bias due to the general tendency to give socially acceptable answers. Nevertheless, these limitations do not diminish the importance of the study in increasing visibility to a relevant problem and the urgency in scaling up actions toward EMTCT in Brazil.

Primary care units and maternity hospitals need to provide easy and effective access to health services to offer quality prenatal and labor care for their clients. It is also important to prepare managers and health professionals to answer clients' needs because only a flexible and personal agenda can improve the services. The way the health care system and certain policy factors are set up can serve as barriers to successful engagement in HIV and syphilis care activities in our fragmented health service. Nonetheless, with robust governmental will and leadership in local communities, opportunities can arise to increase emphasis and improve efforts in this field<sup>(19)</sup>. The success of such efforts depends on overcoming obstacles between actors and integrating surveillance and care, which influence the spectrum of engagement.

### CONCLUSION

The key message from our study is the need to improve communication — among policy makers, managers, health professionals, and the general population. New guidelines have to reach their public — physicians and nurses — in order to be read, accepted, and implemented. Managers have to follow the availability of tests and drugs closely, avoiding shortages that could hamper efforts already in place to tackle syphilis and HIV. Moreover, the general population should be better informed about STIs and their impacts on their lives and the lives of others — and also to be heard by their caregivers, who must be able to answer their questions and help them to find their own answers.

### Participation of each author

All authors were active and had equal participation in the study.

### Funding

This study received support from the Brazilian Ministry of Health, National Health Foundation. Grant # Term 323/2014.

### **Conflict of interests**

The authors declare no conflict of interests.

### **Ethics Committee Approval for Human Research**

The Research Ethics Committee from the Universidade Federal do Espírito Santo approved the project (#640,580/2014).

### REFERENCES

- Victora CG, Barreto ML, do Carmo Leal M, Monteiro CA, Schmidt MI, Paim J, et al. Lancet Brazil Series Working Group. Health conditions and health-policy innovations in Brazil: the way forward. Lancet. 2011;377(9782):2042-53. http://doi.org/10.1016/S0140-6736(11)60055-X
- Wasserheit JN. The Significance and Scope of Reproductive Tract Infections among Third World Women. Int J Gynecol Obstet. 1989;30(Suppl.):145-68. https://doi.org/10.1016/0020-7292(89)90115-X

- Newman L, Kamb M, Hawkes S, Gomez G, Say L, Seuc A, et al. Global estimates of syphilis in pregnancy and associated adverse outcomes: analysis of multinational antenatal surveillance data. PLoS Med. 2013;10(2):e1001396. https://doi.org/10.1371/journal.pmed.1001396
- Macdonald EM, Ng R, Bayoumi AM, Raboud J, Brophy J, Masinde KI, et al. Adverse Neonatal Outcomes Among Women Living With HIV: A Population-Based Study. J Obstet Gynaecol Can. 2015;37(4):302-9. http://doi.org/10.1016/S1701-2163(15)30279-6
- Gardner EM, McLees MP, Steiner JF, Del Rio C, Burman WJ. The spectrum of engagement in HIV care and its relevance to test and-treat strategies for prevention of HIV infection. Clin Infect Dis. 2011;52(6):793-800. https:// doi.org/10.1093/cid/ciq243
- Taylor M, Newman L, Ishikawa N, Laverty M, Hayashi C, Ghidinelli M, et al. Elimination of mother-to-child transmission of HIV and Syphilis (EMTCT): Process, progress, and program integration. PLoS Med. 2017;14(6):e1002329. http://doi.org/10.1371/journal.pmed.1002329
- Kuznik A, Lamorde M, Nyabigambo A, Manabe YC. Antenatal Syphilis Screening Using Point-of-Care Testing in Sub-Saharan African Countries: A Cost Effectiveness Analysis. PLoS Med. 2013;10(11):e1001545. http:// doi.org/10.1371/journal.pmed.1001545
- Kendall T. Consequences of Missed Opportunities for HIV Testing during Pregnancy and Delayed Diagnosis for Mexican Women, Children and Male Partners. PLoS One. 2014;9(11):e109912. http://doi.org/10.1371/ journal.pone.0109912
- UNAIDS. Global plan towards the elimination of new HIV infections among children by 2015 and keeping their mothers alive [Internet]. UNAIDS; 2011 [cited on Jun. 29, 2018]. Available at: http://www.unaids. org/en/resources/documents/2011/name,60876,en.asp
- PAHO. Plan of action for the prevention and control of HIV and sexually transmitted infections 2016-2021. 55th Directing Council, 68th Session of the Regional Committee of WHO for the Americas [Internet]. Washington, D.C.: PAHO; 2016 [cited on Nov. 10, 2018]. Available at: https://www. paho.org/hq/dmdocuments/2016/CD55-14-e.pdf
- Mugavero MJ, Norton WE, Saag MS. Health care system and policy factors influencing engagement in HIV medical care: piecing together the fragments of a fractured health care delivery system. Clin Infect Dis. 2011;52(Suppl. 2):S238-46. http://doi.org/10.1093/cid/ciq048
- Brasil. Ministério da Saúde. Secretaria de Vigilância em Saúde. Departamento de DST, Aids e Hepatites Virais. Boletim Epidemiológico HIV/Aids [Internet]. Brasília: Ministério da Saúde; 2017 [cited on Set. 21, 2019]. Available from: http://www.aids.gov.br/pt-br/pub/2017/boletimepidemiologico-hivaids-2017
- Brasil. Ministério da Saúde. Secretaria de Vigilância em Saúde. Departamento de DST, Aids e Hepatites Virais. Boletim Epidemiológico de sífilis [Internet]. Brasília: Ministério da Saúde; 2017. Available from: http:// www.aids.gov.br/pt-br/pub/2017/boletim-epidemiologico-de-sifilis-2017
- Miranda AE, Pereira GF, Araújo MA, Silveira MF, Tavares L de L, Silva LC, et al. Evaluation of the cascade of care in prevention of mother-to-child HIV transmission in Brazil. Cad Saúde Pública. 2016;32(9):e00118215. http://doi.org/10.1590/0102-311X00118215
- Saraceni V, Pereira GFM, da Silveira MF, Araújo MAL, Miranda AE. Epidemiological surveillance of vertical transmission of syphilis: data from six federal units in Brazil. Rev Panam Salud Publica. 2017;41:e44.
- Skovdal M, Cornish F. Qualitative Research for Development. Rugby, UK: Practical Action Publishing; 2015.
- Cunha FJAP, Lázaro CP, Pinheiro HBB. Conhecimento, inovação e comunicação em serviços de saúde. Rio de Janeiro: Editora Fiocruz; 2014. 240p.
- World Health Organization. A short guide on methods: measuring the impact of national PMTCT programmes: towards the elimination of new HIV infections among children by 2015 and keeping their mothers alive [Internet]. Geneva: World Health Organization; 2012. Available from: http://www.who.int/hiv/pub/mtct/national\_pmtct\_guide/en/
- Caffe S, Perez F, Kamb ML, Gomez Ponce de Leon R, Alonso A, Midy R, et al. Cuba validated as the first country to eliminate mother to child transmission of human immunodeficiency virus and congenital syphilis: Lessons learned from the implementation of the global validation methodology. Sex Transm Dis. 2016;43(12):733-6. https://doi. org/10.1097/OLQ.00000000000528

- World Health Organization. WHO validates elimination of mother-tochild transmission of HIV and syphilis in Thailand, Armenia, Belarus and the Republic of Moldova [Internet]. Geneva: World Health Organization; 2016. Available from: http://www.who.int/reproductivehealth/news/ emtct-hiv-syphilis/en/
- Ishikawa N, Newman L, Taylor MM, Essajee S, Pendse R, Ghidinelli M. Elimination of mother-to-child transmission of HIV and syphilis in Cuba and Thailand. Bull World Health Organ. 2016;94(11):787-787A. https:// dx.doi.org/10.2471%2FBLT.16.185033
- Albuquerque MV, Viana ALA, Lima LD, Ferreira MP, Fusaro ER, Iozzi FL. Desigualdades regionais na saúde: mudanças observadas no Brasil de 2000 a 2016. Ciênc Saúde Coletiva. 2017;22(4):1055-64. https://dx.doi. org/10.1590/1413-81232017224.26862016
- Galvao TF, Silva MT, Serruya SJ, Newman LM, Klausner JD, Pereira MG, et al. Safety of benzathine penicillin for preventing congenital syphilis: a systematic review. PLoS One. 2013;8(2):e56463. http://doi.org/10.1371/ journal.pone.0056463
- 24. Iroezi N, Mindry D, Kawale P, Chikowi G, Jansen P, Hoffman R. A qualitative analysis of the barriers and facilitators to receiving care in a prevention of mother-to-child program in Nkhoma, Malawi. Afr J Reprod Health. 2013;17(4 0 0):118-29.
- Serruya SJ, Cecatti JG, Lago TGO. O Programa de Humanização no Pré-natal e Nascimento do Ministério da Saúde no Brasil: resultados iniciais. Cad Saúde Pública. 2004;20(5):1281-9. http://dx.doi.org/10.1590/ S0102-311X2004000500022
- Domingues RM, Viellas EF, Dias MA, Torres JA, Theme-Filha MM, Gama SG, et al. Adequacy of prenatal care according to maternal characteristics in Brazil. Rev Panam Salud Publica. 2015;37(3):140-7.

- 27. Mendoza-Sassi RA, Cesar JA, Teixeira TP, Ravache C, Araújo GD, Silva TC. Diferenças no processo de atenção ao pré-natal entre unidades da Estratégia Saúde da Família e unidades tradicionais em um município da Região Sul do Brasil. Cad Saúde Pública. 2011;27(4):787-96. http://dx.doi.org/10.1590/S0102-311X2011000400018
- Silveira DS, Santos IS, Costa JSD. Atenção pré-natal na rede básica: uma avaliação da estrutura e do processo. Cad Saúde Pública. 2001;17(1):131-9. http://dx.doi.org/10.1590/S0102-311X2001000100013
- Domingues RM, Lauria L de M, Saraceni V, Leal M do C. Treatment of syphilis during pregnancy: knowledge, practices and attitudes of health care professionals involved in antenatal care of the Unified Health System (SUS) in Rio de Janeiro City. Ciênc Saúde Coletiva. 2013;18(5):1341-51. http://dx.doi.org/10.1590/S1413-81232013000500019

### Address for correspondence: ANGELICA ESPINOSA MIRANDA

Departamento de Medicina Social, Universidade Federal do Espírito Santo Av. Marechal Campos, 1468 – Bairro Maruípe

Vitória (ES), Brazil CEP: 29040-091 E-mail: amiranda.ufes@gmail.com

Received on: 07.09.2019 Approved on: 09.09.2019

# ANTIRETROVIRAL AGENTS AGAINST HUMAN IMMUNODEFICIENCY VIRUS: AN OVERVIEW OF CURRENT DRUGS AND NEW PERSPECTIVES

### Agentes antiretrovirais contra o vírus da imunodeficiência humana: uma visão geral das drogas atuais e novas perspectivas

Ingrid Barcelos de Oliveira<sup>1,2</sup>, Claudio Cesar Cirne-Santos<sup>1,2</sup>, Caroline de Souza Barros<sup>3</sup>, Rosa Teixeira Pinho<sup>4</sup>, Izabel Christina Nunes de Palmer Paixão<sup>1,2</sup>

### ABSTRACT

Introduction: Since its discovery in the 1980s, the human immunodeficiency virus (HIV) has been the target of many studies. Nowadays, estimates show that 36.7 million people are infected with HIV worldwide. In Brazil, HIV infection overcomes 840 thousand people. Globally, only 53% of the HIV infected people are under antiretroviral therapy. Significant advances in antiretroviral therapy have been made since the introduction of zidovudine in 1987. **Objective:** To advance the discoveries of the available antivirals demonstrating their functional specificities. **Methods:** We performed a systematic review with a bibliographic survey in the Index Medicus/MEDLINE and PubMed databases for periodical and indexed articles, from 2013 to 2018 that reported on antiretrovirals used or not in the clinical practice. **Results:** Currently, there are six classes of antiretroviral drugs: nucleoside reverse transcriptase inhibitors (NNRTIs), protease inhibitors (PIs), fusion inhibitors (FIs), entry inhibitors (CCRIs), and HIV integrase strand transfer inhibitors (INIs or INSTIs). In summary, several antiretroviral agents under development make HIV entry, reverse transcription, integration, and maturation emerging drug become targets. **Conclusion:** A multifaceted approach to antiretroviral therapy, using combinations of inhibitors that target different steps of the viral life cycle, has the best potential for long-term control of HIV infection. **Keywords:** AIDS; HIV; antiretroviral therapy, highly active.

### RESUMO

Introdução: Desde sua descoberta na década de 1980, o vírus da imunodeficiência humana (HIV) tem sido alvo de muitos estudos. Atualmente, as estimativas mostram que 36,7 milhões de pessoas estão infectadas pelo HIV em todo o mundo. No Brasil, a infecção pelo HIV supera 840 mil pessoas. Globalmente, apenas 53% das pessoas infectadas pelo HIV estão sob terapia antirretroviral. Avanços significativos na terapia antirretroviral (TARV) foram feitos desde a introdução da zidovudina (AZT) em 1987. **Objetivo:** O objetivo deste estudo foi descrever a descoberta dos antivirais disponíveis atualmente, demonstrando suas especificidades funcionais. **Métodos:** Foi realizada uma revisão sistemática com levantamento bibliográfico nas bases de dados Index Medicus/MEDLINE e PubMed para artigos periódicos e indexados, no período de 2013 a 2018, que relataram antirretrovirais utilizados ou não na prática clínica. **Resultados:** Atualmente, existem seis classes de medicamentos antirretrovirais: inibidores nucleosídeos da transcriptase reversa (NRTIs), inibidores da protease (IPs), inibidores de fusão (FIs), inibidores de entrada (CCRIs) e transferência da cadeia da integração e da maturação, alvos dos medicamentos emergentes. **Conclusão:** Uma abordagem multifacetada para a TAR, usando combinações de inibidores que visam diferentes etapas do ciclo de vida viral, tem o melhor potencial para o controle da infecção pelo HIV a longo prazo. **Palavras-chave:** AIDS, HIV, terapia antirretroviral

### **INTRODUCTION**

HIV is the etiologic agent of acquired immunodeficiency syndrome (AIDS). This syndrome is characterized by a decrease in the total number of T CD4<sup>+</sup> cells, as well as an increase in the incidence of opportunistic infections and progressive failure of the immune system. The first report of AIDS occurred in the early 1980s, with cases of homosexuals with depletion of circulating T lymphocytes in Los Angeles, United States<sup>(1)</sup>.

HIV-1 is a single-stranded RNA virus inserted on the *Retroviridae* family. Its particle possesses 100 to 150 nanometers of diameter with

a lipid envelope that contains spike-form glycoproteins<sup>(2,3)</sup>. HIV can infect some immune system cells that express CD4 receptors, such as CD4<sup>+</sup> T helper cells, macrophages, and dendritic cells. Typically, HIV-1 infection is characterized by an initial phase of high-level viremia, followed by a long period of persistent virus replication at a lower level. Viral persistence occurs despite specific antiviral immune responses, which include the generation of neutralizing antibodies<sup>(4)</sup>. The progression of HIV infection can be divided into three phases: acute or primary, persistent chronic phase, and AIDS. The markers of AIDS progression, which clinically define these steps, are the CD4<sup>+</sup> T cell count, the plasma viral load, and the presence or absence of clinical manifestations<sup>(5)</sup>.

Three decades and a half after its discovery and recognition, HIV has emerged as a global epidemic present in all continents, despite the massive campaigns of prevention and growing treatment strategies. The 2017 statistics showed that there were about 36.7 million people living with HIV in 2016<sup>(6)</sup>. Since 1980 until 2016, 842,710 people have been HIV infected in Brazil, and in terms of access to antiretroviral therapy, a total of 495.275 individuals are under treatment<sup>(7)</sup>. Studies have indicated that morbidity and mortality from HIV

 <sup>&</sup>lt;sup>1</sup>Post-Graduation in Science and Biotechnology Program, Institute of Biology, Universidade Federal Fluminense – Niterói (RJ), Brazil.
 <sup>2</sup>Laboratory of Molecular Virology, Department of Cellular and Molecular Biology, Universidade Federal Fluminense – Niterói (RJ), Brazil.
 <sup>3</sup>Department of Immunobiology, Universidade Federal Fluminense – Niterói (RJ), Brazil.

<sup>&</sup>lt;sup>4</sup>Laboratory of Clinical Immunology, Department of Immunology, Instituto Oswaldo Cruz – Rio de Janeiro (RJ), Brazil.

infection have decreased since the introduction of ART. Although the decrease in statistics is evident, the lack of effective treatment and/or protective vaccine and strategies capable of inhibiting the virus replication are still the significant challenges of scientists<sup>(8,9)</sup>.

### **OBJECTIVE**

The aim of this study is to address the importance of less toxic compounds with higher antiviral activity that can reduce the morbidity and mortality of HIV infection. In addition, it aimed to produce a correlation between azidothymidine (AZT) and other drugs that even though are available for clinical use, have high levels of toxicity.

### **METHODS**

We conducted a research in Index Medicus/MEDLINE and PubMed databases on articles that reported the currently available antiretrovirals. For this analysis, we used the keywords "AIDS", "HIV" and "antiretroviral therapy" in the database to find articles from the last five years (from 2013 and 648 articles), evaluated and followed our exclusion criteria that are articles that fully descriptively address compounds in clinical use or compounds that are in the preclinical (experimental) phase. In addition, the compounds have established toxicity and mechanism of action data. Therefore, only 38 articles were selected to compose this paper.

### RESULTS

Forty-five articles from indexed journals and 38 selected for this review work were evaluated. We observed that since the discovery of AZT as the first drug available for use in the HIV infection treatment that shows great results, several substances have been studied and made available to make up the current treatment, which has been very effective.

### DISCUSSION

### Antiretroviral therapy

After 35 years from HIV discovery, many studies have been done and achieved for creating anti-HIV drugs. There is still no cure for such disease; therefore, once the highly active antiretroviral therapy (HAART) — the treatment currently used for HIV seropositive individuals — has been initiated, it must be continued throughout the patient's life<sup>(10)</sup>. The HIV-1 reverse transcriptase (RT), protease (PR) and integrase (IN) enzymes, as well as the main steps of the replicative cycle, are the main targets of antiretroviral therapy.

HAART consists of a combination of drugs that aim to inhibit enzymes and/or fusion/entry mechanisms of the virus into the cell during the steps of the HIV replicative cycle. A remarkable effect of HAART is the regeneration of CD4<sup>+</sup> and CD8<sup>+</sup> T cells and re-establishment of the immune function stably at long term. Therapy acts to reduce viral load in blood plasma, promoting higher survival to individuals<sup>(11-13)</sup>. Initially, the use of two nucleoside RT inhibitors (NRTI) and one non-nucleoside RT inhibitor (NNRTI) or two NRTIs and a third drug, either an integrase inhibitor or the protease inhibitor as HAART standard approach, is recommended for the treatment of AIDS<sup>(14,15)</sup>.

There are 28 available drugs approved by the U.S. Food and Drug Administration (FDA) for the treatment of HIV infection, including: eight nucleoside reverse transcriptase inhibitors (NRTIs); five non-nucleoside reverse transcriptase inhibitors (NNRTIs); ten protease inhibitors (PIs); one fusion inhibitor (FI); one entry inhibitor (CCRI); three integrase inhibitors (INIs or INSTIs), as see in **Table 1**<sup>(16-18)</sup>.

Despite the continued developments in AIDS treatment and virus combat, infected individuals remain susceptible to viral and bacterial infections, such as influenza and pneumococcus, which can raise the level of morbidity and mortality. Therefore, alternative strategies and new targets need to be considered to combat the increasing AIDS pandemic<sup>(19)</sup>.

**Table 1 –** Antiretroviral drugs currently used in the HIV infection treatment.

Drug Class	Generic name	Year of
Mode of action	Seneric name	approval
	zidovudine (ZDV,	1987
Nucleoside reverse transcriptase inhibitors (NRTIs) They are phosphorylated by cellular enzymes; competitively inhibit viral DNA synthesis or cause DNA chain termination	azidothymidine, AZT)	1007
	didanosine (ddl,	1991
	dideoxyinosine)	1001
	zalcitabine (ddC,	1002
	dideoxycytidine)	1992
	stavudine (d4T)	1994
	lamivudine (3TC)	1995
	abacavir (ABC)	1998
	tenofovir (TDF)	2001
	emtricitabine (FTC)	2003
Non-nucleoside reverse	nevirapine (NVP)	1996
transcriptase inhibitors	delavirdine (DLV)	1997
(NNRTIS)	efavirenz (EFV)	1998
They are non-competitive	etravirine (ETR)	2008
inhibitors of viral DNA		
synthesis; directly connect the	rilpivirine (RPV)	2011
enzyme		
	saquinavir (SQV)	1995
	indinavir (IDV)	1996
	ritonavir (RTV)	1996
Protease inhibitors (PIs)	nelfinavir (NFV)	1997
They bind to the active site of	amprenavir (APV)	1999
protease, thereby inhibiting	lopinavir (LPV)	2000
enzyme function	fosamprenavir (FOS-APV)	2003
	atazanavir (ATV)	2003
	tipranavir (TPV)	2005
	darunavir (DRV)	2006
Fusion inhibitors (FI)		
They bind to HR1 region of	enfuvirtide (T-20)	2003
envelope glycoprotein gp41		
Entry inhibitors – CCR5 co-		
receptor antagonist (CCRI)		
They inhibit the interaction	maraviroc (MVC)	2007
of glycoprotein gp120 with		
CCR5 chemokine receptor		
HIV integrase strand transfer	raltegravir (RAL)	2007
inhibitors (INI or INSTI)	elvitegarvir (EVG)	2012
They block integrase,		
preventing the insertion of	dulotegravir (DTG)	2013
viral DNA into the host DNA		

Source: adapted from Pau and George<sup>(16)</sup>, da Cunha, Maselli<sup>(17)</sup>, FDA<sup>(18)</sup>.

### Pharmacokinetic enhancer

Most of HIV protease inhibitors need a pharmacokinetic enhancer to assure their efficiency. Although ritonavir is a protease inhibitor, it is commonly used as a pharmacokinetic enhancer. Due to high daily doses and some side effects related to ritonavir use, the administration of cobicistat has become more frequent. In addition, cobicistat (COBI) inhibits cytochrome CYP3A more selectively than ritonavir<sup>(20)</sup>.

COBI is a pharmacokinetic enhancer (pharmacoenhancer) that inhibits the metabolism of the atazanavir and darunavir antiretrovirals HIV-1 protease inhibitors (PIs) and is known for prolonging their effects. It was first approved by the US Food and Drug Administration (FDA) in 2012 combined with other drugs (elvitegravir/emtricitabine/ tenofovir). Then, its single use was approved in 2014. More recently, in 2015, its use combined with atazanavir (ATV/COBI) and darunavir (DRV/COBI) was approved for use in treatment-naive and treatment-experienced adults in combination with other ARV agents<sup>(21)</sup>.

COBI possesses a structure analog to ritonavir. Synthesized from ritonavir, it is an inhibitor of human cytochrome P-450 (CYP) 3A and is used to increase the systemic exposure of antiretrovirals metabolized by CYP3A enzymes, in other words, it helps to optimize the activity of these antiretrovirals<sup>(20,21)</sup>.

COBI inactivates CYP3A like ritonavir, in a concentration- and time-dependent manner. However, unlike ritonavir, COBI has no anti-HIV activity, which is essential to develop no resistance to antiretrovirals. Moreover, it does not disrupt the antiviral activity of PIs and several other ARVs against HIV. As a nonselective CYP3A inhibitor, the use of ritonavir becomes difficult in patients receiving multiple medications. COBI has demonstrated high efficacy. It can be compared and is considered non-inferior to standard RTV as a pharmacoenhancer. Hence, COBI offers an alternative to ritonavir boosting, with efficacy and safety in treatment-naive and -experienced adults<sup>(22,23)</sup>.

### Microbicides

Microbicides are experimental products containing drugs that could be inserted into the vagina or rectum to safely block sexual transmission of HIV. A microbicide would deliver an anti-HIV agent to the mucous membranes, lining the surface of the vagina or rectum through a substance, such as a gel, intravaginal ring, or film. A safe, effective, desirable, and affordable microbicide against HIV could help to prevent many new infections. Microbicide development is a challenge due to many factors, such as: no precedent that guarantees the strategy success, difficulty of achieving secure therapeutic levels, lack of markers, among others<sup>(24)</sup>.

There are some essential characteristics for prioritizing microbicide development. They are safety, efficacy, cost, acceptability, appropriate drug delivery, long-term effectiveness, potential for resistance and impact on therapy, prioritization of best-in-class products<sup>(25)</sup>.

Women are two to four times more susceptible than men to acquire sexually transmitted HIV infection, and recent research from the Joint United Nations Programme on HIV/AIDS (UNAIDS) show that 51% of people living with HIV worldwide are women. Moreover, AIDS-related illnesses remain the leading cause of death among women at reproductive age (15–49 years) in the world<sup>(26,27)</sup>.

Microbicides currently used for pre-exposure prophylaxis strategy to prevent the sexual transmission of HIV are astodimer, dapivirine, and tenofovir<sup>(28)</sup>.

### Astodimer

Astodimer or VivalGel is a topical microbicide that is applied to the vagina, rectum, or used in condoms during a sexual intercourse. Studies have demonstrated the antiviral and antibacterial activities of the active principle. Astodimer belongs to a class of drugs that inhibits virus entry into the cell through polyanion, electrostatically creating an environment that prevents adsorption<sup>(25,29)</sup>.

### Dapivirine

Dapivirine is a non-nucleoside reverse transcriptase inhibitor that prevents the conversion of viral RNA to pro-viral DNA. It is also used as a topic pre-exposure prevention. One study evaluated the distribution of drug concentration of vaginal canal of women who used a silicone ring containing 25 mg of Dapivirine. The concentration was higher in the entire length of ectocervix than in endocervix<sup>(28)</sup>.

### Tenofovir

Tenofovir is a nucleoside reverse transcriptase inhibitor that was developed as a pre-exposure prophylaxis. Tenofovir clinical trials have shown promising results, with low anal and vaginal mucosal irritability. A study evaluated the use of 1% tenofovir vaginal gel for 30 months. There was an increase in cases of mild and self-limiting diarrhea, except for the group receiving the gel compared to the placebo group<sup>(30)</sup>.

## MARINE NATURAL PRODUCTS IN THE REPLICATION OR INFECTIVITY OF HIV AS A CANDIDATE FOR NEW ANTIRETROVIRALS

Marine organisms, as well as their derivatives, are sources of biologically active compounds. The Brazilian coast has a great marine biodiversity, with a broad range of resource of compounds with potential anti-HIV-1 activity<sup>(31)</sup>. Researchers' interest in investigating marine natural products is because the current antiretroviral therapy provides an emergence increase of drug resistance. Thus, it is important for the identification of new drugs with inhibitory action of both reverse transcriptase and other viral enzymes. Literature has demonstrated that the compounds have promising antiviral activity.

Studies have shown many compounds with rich chemical diversity and high biological activity against different microorganisms, especially HIV. A study of 257 pure compounds showed that at least four had an interesting potential to induce the production of latent proviruses, which may be an interesting combination strategy and may generate a more efficient low-dose combined effect<sup>(32)</sup>. Other studies with marine compounds had high repercussion and showed significant results, as flavonoids isolated from Caribbean seagrass acting as integrase inhibitors<sup>(33)</sup>. Nanotechnology active compounds with superior pharmacokinetics for site-specific delivery appear to be an interesting strategy for the use in more deadly diseases, such as cancer, HIV/AIDS, and neurological disorders. However, a large number of active compounds against HIV can be removed from the marine environment, such as polysaccharides, lectins, Aspernigrins with anti-HIV-1 activities and Manzamine, although the best strategy is possibly for compounds that act directly on the virus<sup>(34)</sup>.

In 2004, Pereira and Leão-Ferreira<sup>(35)</sup> demonstrated antiviral activity of two diterpenes isolated denominated hydroxidictiodial and dictiodial from brown alga Dictyota menstrualis against HIV-1. Values between 40 and 70  $\mu$ M of EC<sub>50</sub> in PM-1 cells were obtained. The diterpenes also inhibited the synthesis of pro-viral DNA, and reverse transcriptase was inhibited in a dose-dependent manner with the maximal inhibitory effect varying from 70 to 95% by both compounds. Cirne-Santos et al.<sup>(36)</sup> showed that diterpene 8,10,18-trihydroxy-2, 6-dolabelladiene (dolabelladienetriol) isolated from marine brown alga Dictyota pfaffii inhibits the HIV-1 infection. Dolabelladienetriol was able to inhibit the activity of RT enzyme in a dose-dependent manner, reaching the value of 16.5  $\mu$ M of IC<sub>so</sub>. Infected peripheral blood mononuclear cells (PBMCs) and macrophages presented EC<sub>50</sub> of 8.4 and 1.85  $\mu$ M, respectively. These results show that the diterpene can be a candidate to a new antiretroviral, including for viral reservoirs.

After those and other results, Paixão<sup>(37)</sup> proposed the formulation of a microbicide gel for women with active principles obtained from isolated products of *Dictyota menstrualis* and *Dictyota pfaffii* algae, which act inside the cell preventing HIV replication and inhibiting enzymes of virus life cycle<sup>(37)</sup>.

In 2014, Pardo-Vargas et al.<sup>(38)</sup> reported three new dolabellane diterpenes isolated from Brazilian brown alga *Dictyota pfaffii* with antiviral activity against HIV-1 in MT-2 cells. They presented high values of  $CC_{50}$ , varying between 1345 and 1,456  $\mu$ M. Compounds also inhibited 52, 69 and 83% of the HIV-1 replication, suggesting that these diterpenes could be considered potential new agents for HIV-1 therapy.

Isolated compounds of another brown alga, *Canistrocarpus cervicornis*, demonstrated inhibitory activity against HIV-1 replication and low cytotoxic effect in cell culture. Marine dolasthanes and secodolastane diterpenes inhibit virus replication in a dose-dependent manner. Diterpenes also showed excellent performance to reduce the infective capacity of HIV, which suggests the existence of a mechanism that prevents virus adsorption in the host cell. The promising results of algae extracts in combating HIV-1 replication and its potential virucidal effect require further studies, such as pharmacodynamic tests and *in vitro* and *in vivo* models<sup>(39)</sup>.

Nogueira et al.<sup>(40)</sup> demonstrated the antiviral effect of extracts from red alga *Acanthophora spicifera* on HIV-1 replication. PBMCs were used to evaluate the cytotoxicity ( $CC_{50}$ ) of dichloromethane, ethyl acetate, acetone, and methanol extracts. The respective values were: 31, 45, 38, and 179 µg/mL. The antiviral activity showed that ethyl acetate extract inhibited 60% of viral replication in infected cells, whereas the maximum inhibition was 79% for RT. The results evidence the promising use of *A. spicifera*, especially as a RT inhibitor.

Pioneer studies on the potential antiviral to develop microbicides as a preventive and topical use in genital mucosa are being developed, mainly the use of explant technology due to its easy study of the human mucosal tissue. Stephens et al.<sup>(41)</sup> determined the pretreatment effect with the compound extracted from brown alga *Dictyota pfaffii* dolabelladienetriol in inhibition of HIV-1 replication, and the protective effect of this compound in *ex vivo* explant model. The compound showed high inhibitory activity in PBMCs, achieving from 60 to 80%, and prevented HIV-1 replication macrophages. Tests with explant model (stratified squamous tissue) confirmed the low cytotoxicity of dolabelladienetriol and inhibited viral replication from 21 to 95%, maintaining the viability of the tissue, producing a protective effect on it. These findings indicate the need for further studies to the development of new antiretrovirals.

### CONCLUSIONS

The current available antiretroviral therapy is composed of highly effective drugs that have improved morbidity and mortality of infected individuals, although many have significant toxic effects. The search for new antiretrovirals that have high activity and low toxicity is undoubtedly an important strategy to increase the arsenal of drugs for effective treatment and control of HIV infection. This fact can still be fundamental, especially considering that there are no vaccines available for HIV infection.

### Participation of each author

Ingrid Barcelos de Oliveira, Doctoral student, worked on writing the review article. Claudio Cesar Cirne-Santos is the co-advisor and worked on article review. Caroline de Souza Barros worked on the article review. Rosa Teixeira Pinho worked on the article review. Izabel Christina Nunes de Palmer Paixão is project coordinator.

### Funding

The authors are grateful to *Conselho Nacional de Desenvolvimento Científico e Tecnológico* (CNPq) for their financial support and Productivity Fellowships to I.C.N.P.P. (443930/2014-7 and 304070/2014-9). I.C.N.P.P. (E-26/201.442/2014) also thanks the *Fundação de Amparo à Pesquisa do Estado do Rio de Janeiro* (FAPERJ) for the *Cientista do Nosso Estado* Fellowship. C.C.C.S. thanks to Coordenação de Aperfeiçoamento de Pessoal de Nível Superior (CAPES) for the Postdoc fellowship, and C.S.B. thanks FAPERJ for the Postdoc fellowship (E-26/201.344/2016), in the Postgraduate Program in Sciences and Biotechnology of *Universidade Federal Fluminense* (PPBI-UFF).

### **Conflict of interests**

The authors declare no conflict of interests.

### REFERENCES

- De Cock KM, Jaffe HW, Curran JW. The evolving epidemiology of HIV/AIDS. Aids. 2012;26(10):1205-13. https://doi.org/10.1097/QAD.0b013e328354622a
- Tagliamonte M, Tornesello ML, Buonaguro FM, Buonaguro L. Conformational HIV-1 envelope on particulate structures: a tool for chemokine coreceptor binding studies. J Transl Med. 2011;9(Suppl. 1):S1. https://dx.doi.org/10.1186%2F1479-5876-9-S1-S1
- Collier LKP, Oxford J. Retroviruses HIV-1 and -2 and HTLV. In: Press OU, editor. Human Virology. 5<sup>th</sup> edition. 2016. p. 273-87.
- Wyatt R, Sodroski J. The HIV-1 envelope glycoproteins: fusogens, antigens, and immunogens. Science. 1998;280(5371):1884-8. https://doi. org/10.1126/science.280.5371.1884
- Coffin J, Swanstrom R. HIV pathogenesis: dynamics and genetics of viral populations and infected cells. Cold Spring Harb Perspect Med. 2013;3(1):a012526. https://dx.doi.org/10.1101%2Fcshperspect.a012526

- Akbari M, Fararouei M, Haghdoost AA, Gouya MM, Kazerooni PA. Survival and associated factors among people living with HIV/AIDS: A 30-year national survey in Iran. J Res Med Sci. 2019;24:5. https://dx.doi. org/10.4103%2Fjrms.JRMS\_630\_18
- Brasil. Ministério da Saúde. Boletim Epidemiológico HIV/AIDS. Brasília: Ministério da Saúde, Secretaria de Vigilância em Saúde, PN de DST e AIDS; 2017. Ano V, n. 1.
- Cardoso SW, Luz PM, Velasque L, Torres T, Coelho L, Freedberg KA, et al. Effectiveness of first-line antiretroviral therapy in the IPEC cohort, Rio de Janeiro, Brazil. AIDS Res Ther. 2014;11:29. https://doi. org/10.1186/1742-6405-11-29
- Pang X, Liu Z, Zhai G. Advances in non-peptidomimetic HIV protease inhibitors. Curr Med Chem. 2014;21(17):1997-2011. https://doi.org/10.21 74/0929867321666140217115951
- Sarafianos SG, Marchand B, Das K, Himmel DM, Parniak MA, Hughes SH, et al. Structure and function of HIV-1 reverse transcriptase: molecular mechanisms of polymerization and inhibition. J Mol Biol. 2009;385(3):693-713. https://doi.org/10.1016/j.jmb.2008.10.071
- Agrawal L, Lu X, Jin Q, Alkhatib G. Anti-HIV therapy: current and future directions. Curr Pharm Des. 2006;12(16):2031-55. https://doi. org/10.2174/138161206777442100
- Ramkumar K, Serrao E, Odde S, Neamati N. HIV-1 integrase inhibitors: 2007–2008 update. Med Res Rev. 2010;30(6):890-954. https://doi. org/10.1002/med.20194
- De Luca L, Gitto R, Christ F, Ferro S, De Grazia S, Morreale F, et al. 4-[1-(4-Fluorobenzyl)-4-hydroxy-1H-indol-3-yl]-2-hydroxy-4-oxobut-2-enoic acid as a prototype to develop dual inhibitors of HIV-1 integration process. Antiviral Res. 2011;92(1):102-7. https://doi.org/10.1016/j.antiviral.2011.07.005
- Nanfack AJ, Agyingi L, Noubiap JJN, Ngai JN, Colizzi V, Nyambi PN. Use of amplification refractory mutation system PCR assay as a simple and effective tool to detect HIV-1 drug resistance mutations. J Clin Microbiol. 2015;53(5):1662-71. https://doi.org/10.1128/JCM.00114-15
- Murray AJ, Kwon KJ, Farber DL, Siliciano RF. The latent reservoir for HIV-1: how immunologic memory and clonal expansion contribute to HIV-1 persistence. J Immunol. 2016;197(2):407-17. https://doi. org/10.4049/jimmunol.1600343
- Pau AK, George JM. Antiretroviral therapy: current drugs. Infect Dis Clin North Am. 2014;28(3):371-402. https://doi.org/10.1016/j.idc.2014.06.001
- da Cunha J, Maselli LMF, Stern ACB, Spada C, Bydlowski SP. Impact of antiretroviral therapy on lipid metabolism of human immunodeficiency virus-infected patients: Old and new drugs. World J Virol. 2015;4(2):56-77. https://doi.org/10.5501/wjv.v4.i2.56
- FDA. Approved HIV Medicines [Internet]. FDA; 2019 [cited on Mar. 21, 2019]. Available from: https://aidsinfo.nih.gov/understanding-hiv-aids/ fact-sheets/21/58/fda-approved-hiv-medicines
- Cribbs SK, Lennox J, Caliendo AM, Brown LA, Guidot DM. Healthy HIV-1-infected individuals on highly active antiretroviral therapy harbor HIV-1 in their alveolar macrophages. AIDS Res Hum Retroviruses. 2015;31(1):64-70. https://doi.org/10.1089/AID.2014.0133
- Kakuda TN, Crauwels H, Opsomer M, Tomaka F, van de Casteele T, Vanveggel S, et al. Darunavir/cobicistat once daily for the treatment of HIV. Expert Rev Anti Infect Ther. 2015;13(6):691-704. https://doi.org/10. 1586/14787210.2015.1033400
- Sherman EM, Worley MV, Unger NR, Gauthier TP, Schafer JJ. Cobicistat: review of a pharmacokinetic enhancer for HIV infection. Clin Ther. 2015;37(9):1876-93. https://doi.org/10.1016/j.clinthera.2015.07.022
- Gallant JE, Koenig E, Andrade-Villanueva J, Chetchotisakd P, DeJesus E, Antunes F, et al. Cobicistat versus ritonavir as a pharmacoenhancer of atazanavir plus emtricitabine/tenofovir disoproxil fumarate in treatmentnaive HIV type 1–infected patients: week 48 results. J Infect Dis. 2013;208(1):32-9. https://doi.org/10.1093/infdis/jit122
- Deeks ED. Cobicistat: a review of its use as a pharmacokinetic enhancer of atazanavir and darunavir in patients with HIV-1 infection. Drugs. 2014;74(2):195-206. https://doi.org/10.1007/s40265-013-0160-x
- Singh K, Flores J, Kirby K, Neogi U, Sonnerborg A, Hachiya A, et al. Drug resistance in non-B subtype HIV-1: impact of HIV-1 reverse transcriptase inhibitors. Viruses. 2014;6(9):3535-62. https://doi.org/10.3390/v6093535

- Shattock RJ, Rosenberg Z. Microbicides: topical prevention against HIV. Cold Spring Harb Perspect Med. 2012;2(2):a007385. https://dx.doi. org/10.1101%2Fcshperspect.a007385
- Wang L, Wang L, Smith KM, Li L-M, Ming S, Lü J, et al. Heterosexual transmission of HIV and related risk factors among serodiscordant couples in Henan province, China. Chin Med J (Engl). 2013;126(19):3694-700.
- 27. UNAIDS. Global AIDS update 2017. Geneva: UNAIDS; 2017. 198p.
- Nel A, Haazen W, Nuttall J, Romano J, Rosenberg Z, van Niekerk N. A safety and pharmacokinetic trial assessing delivery of dapivirine from a vaginal ring in healthy women. Aids. 2014;28(10):1479-87. https://doi. org/10.1097/QAD.00000000000280
- Fletcher PS, Wallace GS, Mesquita PM, Shattock RJ. Candidate polyanion microbicides inhibit HIV-1 infection and dissemination pathways in human cervical explants. Retrovirology. 2006;3:46. https://doi.org/10.1186/1742-4690-3-46
- Karim QA, Karim SSA, Frohlich JA, Grobler AC, Baxter C, Mansoor LE, et al. Effectiveness and safety of tenofovir gel, an antiretroviral microbicide, for the prevention of HIV infection in women. Science. 2010;329(5996):1168-74. https://doi.org/10.1126/science.1193748
- Berlinck RG, Hajdu E, da Rocha RM, de Oliveira JH, Hernández IL, Seleghim MH, et al. Challenges and rewards of research in marine natural products chemistry in Brazil. J Nat Prod. 2004;67(3):510-22. https://doi. org/10.1021/np0304316
- Richard K, Williams D, de Silva E, Brockman M, Brumme Z, Andersen R, et al. Identification of novel HIV-1 latency-reversing agents from a library of marine natural products. Viruses. 2018;10(7). https://doi.org/10.3390/v10070348
- Martins BT, Correia da Silva M, Pinto M, Cidade H, Kijjoa A. Marine natural flavonoids: chemistry and biological activities. Nat Prod Res. 2019;33(22):3260-72. https://doi.org/10.1080/14786419.2018.1470514
- Parimala S, Begum A. A review on anti hiv agents from marine sources. World J Pharm Res. 2019;8(6).
- Pereira H, Leão-Ferreira L, Moussatché N, Teixeira V, Cavalcanti D, Costa L, et al. Antiviral activity of diterpenes isolated from the Brazilian marine alga Dictyota menstrualis against human immunodeficiency virus type 1 (HIV-1). Antiviral Res. 2004;64(1):69-76. https://doi.org/10.1016/j.antiviral.2004.06.006
- Cirne-Santos CC, Teixeira VL, Castello-Branco LR, Frugulhetti IC, Bou-Habib DC. Inhibition of HIV-1 replication in human primary cells by a dolabellane diterpene isolated from the marine algae Dictyota pfaffii. Planta Med. 2006;72(4):295-9. https://doi.org/10.1055/s-2005-916209
- Paixão TV. A Cura que Vem do Mar (The Cure that Comes from the Sea). DST - Jornal Brasileiro de Doenças Sexualmente Transmissíveis. 2008.
- Pardo-Vargas A, de Barcelos Oliveira I, Stephens P, Cirne-Santos C, de Palmer Paixão I, Ramos F, et al. Dolabelladienols A–C, new diterpenes isolated from Brazilian brown alga Dictyota pfaffii. Mar Drugs. 2014;12(7):4247-59. https://dx.doi.org/10.3390%2Fmd12074247
- de Souza Barros C, Cirne-Santos CC, Garrido V, Barcelos I, Stephens PRS, Giongo V, et al. Anti-HIV-1 activity of compounds derived from marine alga Canistrocarpus cervicornis. J Appl Phycol. 2016;28:2523-7. http://doi.org/10.1007/s10811-015-0776-1
- Nogueira CCR, Paixão ICNP, Santos CCC, Stephens PRS, Villaça RC, Pereira HS, et al. Anti-HIV-1 activity in human primary cells and Anti-HIV-1 RT inhibitory activity of extracts from the red seaweed Acanthophora spicifera. J Med Plants Res. 2016;10(35):621-5. http://doi. org/10.5897/JMPR2016.6208
- Stephens PRS, Cirne-Santos CC, de Souza Barros C, Teixeira VL, Carneiro LAD, Amorim LSC, et al. Diterpene from marine brown alga Dictyota friabilis as a potential microbicide against HIV-1 in tissue explants. J Appl Phycol. 2017;29(2):775-80. http://doi.org/10.1007/s10811-016-0925-1

### Address for correspondence:

*IZABEL CHRISTINA NUNES DE PALMER PAIXÃO* Outeiro de São João Batista, s/n, Campus do Valonguinho – Centro Niterói (RJ), Brazil CEP: 24020-140 E-mail: izabeluff@gmail.com

Received on: 05.10.2019 Approved on: 07.11.2019

# **RECURRENT VULVAR CONDILOMA TREATMENT IN INFANT: A CASE REPORT**

### TRATAMENTO DE CONDILOMA VULVAR RECORRENTE EM LACTENTE: UM RELATO DE CASO

Denise Gasparetti Drumond<sup>1</sup>, Gabriel Duque Pannain<sup>1</sup>, Tarlline Ribeiro Sirilo<sup>1</sup>, Miralva Aurora Galvão Carvalho<sup>1</sup>, Samuel Drumond Esperança<sup>2</sup>, Homero Gonçalves Junior<sup>1</sup>

### ABSTRACT

Introduction: Anogenital warts are one of the major clinical manifestations of the Human Papillomavirus (HPV). Case reports in children have grown in the last decades; however, there are still difficulties in determining the virus epidemiology and the best therapy for this age group. Objective: To report a case of recurrent vulvar condyloma in a sexually abused infant who presented complete resolution of the lesions after the use of topical imiquimod. Methods: Data research and medical record review were performed in addition to a qualitative study consisted of a wide literature appreciation on the subject. Results: After undergoing extensive therapy, the patient was successfully treated using topical imiquimod. Conclusion: Observation of this particular case suggests that imiquimod may be a safe and effective therapeutic alternative for the treatment of condyloma in the pediatric population. However, theoretical foundations for such conduct are scarce. Thus, the need for further studies on the subject is reinforced. Keywords: Condylomata acuminata; Papillomaviridae; vulvar diseases; imiquimod.

### RESUMO

Introdução: As verrugas anogenitais constituem uma das principais manifestações clínicas da infecção pelo vírus do Papiloma Humano (HPV). Relatos de casos em crianças têm crescido nas últimas décadas; contudo, ainda há dificuldades em determinar a epidemiologia do vírus e definir a melhor terapêutica para essa faixa etária. Objetivo: Relatar um caso de condiloma vulvar recorrente em uma lactente abusada sexualmente que apresentou resolução completa das lesões após o uso de imiquimode tópico. Métodos: Foi realizado levantamento de dados e revisão de prontuário, além de estudo qualitativo composto por apreciação ampla da literatura acerca do tema em questão. Resultados: Depois de ser submetido à extensa terapêutica, a paciente foi tratada com sucesso utilizando imiquimode tópico. Conclusão: A observação desse caso específico sugere que o imiquimode pode ser uma alternativa terapêutica segura e eficaz para o tratamento de condiloma na população pediátrica. Entretanto, embasamentos teóricos para tal conduta são escassos. Assim, reforça-se a necessidade do desenvolvimento de mais estudos sobre a temática.

Palavras-chave: Condiloma acuminado; HPV; doenças da vulva; imiquimode.

### **INTRODUCTION**

Anogenital warts are one of the main clinical manifestations of Human Papillomavirus (HPV) infection<sup>(1,2)</sup>. According to the medical literature, cases reported in pediatric age have grown in recent decades<sup>(2,3)</sup>. However, determining the epidemiology of the virus and defining the best therapy for the lesions in this age group is still a major dilemma in Pediatrics<sup>(4,5)</sup>.

Studies indicate that the period between the initial infection and the emergence of clinical manifestations of the virus vary from three weeks to eight months, except in children, as the period is still unknown<sup>(2)</sup>. In adulthood, the virus is transmitted primarily via sexual intercourse, while the main forms of HPV transmission in children, in addition to sexual transmission (either oral-genital, genital-genital, genital-anal contact or digital manipulation of the children's intimate parts), include vertical transmission, self-inoculation and direct or indirect personal contact with contaminated objects or surfaces<sup>(4,6-9)</sup>.

The prevalence of HPV in sexually abused children varies from 5 to 33%, and around 16% in children without suspicion of abuse<sup>(5)</sup>.

The incidence of sexual abuse is difficult to estimate, mainly because a large percentage of these cases goes unnoticed<sup>(8)</sup>. In addition, this condition is associated with high rates of adverse outcomes, with physical and psychological consequences for the child<sup>(8)</sup>. In regard to the implications of a sexual abuse investigation, other forms of contagion should always be considered before transmission is defined as occurred by sexual abuse<sup>(7)</sup>.

Several treatment options are available for genital warts, but few of them are adequate for children<sup>(10)</sup>. The selected therapy should then be individualized, considering mainly the child's tolerability<sup>(7)</sup>.

According to the above mentioned, registering a case of recurrent vulvar condyloma in infant is proposed, in order to illustrate the difficulties found for its therapeutic approach.

### **CASE REPORT**

The case report was approved by the local Ethics and Research Committee with Opinion No. 3.370.374.

A nine-month-old child admitted to the University Hospital of Universidade Federal de Juiz de Fora (UFJF) with vulvar condylomatosis (**Figure 1**), already under the care of the Tutelary Council, was referred to the surgery center for lesion excision. Cauterization of the lesions with an electric scalpel was the primary option. The child was discharged three days after the procedure with recommendation of local care. After approximately 45 days of cauterization,

<sup>&</sup>lt;sup>1</sup>Department of Surgery, Medicine School, Universidade Federal de Juiz de Fora – Juiz de Fora (MG), Brazil.

<sup>&</sup>lt;sup>2</sup>Medical Sciences and Health School, Medicine School, Universidade Federal de Juiz de Fora – Juiz de Fora (MG), Brazil.

the patient returned to the outpatient clinic with recurrence of the lesions (**Figure 2**).

After resurgence of the condylomatous lesions, weekly cauterizations with 70% trichloroacetic acid were elected as treatment. The procedure was performed for two months, but there was still recurrence of the symptoms (**Figure 3**).

Due to the systematic recurrence of the lesions, the treatment with *Thuya occidentallis* was chosen. After 30 days of application of a creamy formulation of *Thuya Occidentalis* (three times a week), the condylomatous lesions disappeared. However, the lesions reappeared after two months. Then, using Imiquimod for three weeks (three times a week) in a residual condyloma, a definite improvement of the lesions was finally observed (**Figure 4**).

### DISCUSSION

The way infants acquire condyloma should be a priority in medical care. It is important to determine whether there is any evidence of sexual abuse and try to establish the origin of the virus. Faced with situations such as the suspicion of sexual abuse reported in this work, it is also essential that care is provided by a multidisciplinary team, ensuring a complete medical and social assessment<sup>(7,11)</sup>.

Suspicion of abuse should be raised regarding the following facts: the explanations about the lesion are vague or absent, there are different versions of the fact, and when the story is inconsistent with the physical findings<sup>(12,13)</sup>. In this context, the case reported here



**Figure 1** – Multiple vegetative lesions conflowing in the perineal and perianal region, and perilesional erythema present in the first care, prior to any treatment.



**Figure 2 –** Recurrence of vegetative lesions 45 days after resection with electric scalpel.



**Figure 3** – Papular lesions, in lower volume, relapsed in the perineal and perianal region after treatment with 70% trichloroacetic acid.

was submitted to a multidisciplinary team, communicated to the Child Protection Council, and after the sexual abuse by the father was confirmed, the Tutelary Council carried on with the legal-assistance measures.

Regarding the treatment of anogenital lesions in pediatric age, it is known that studies with this approach are still limited, with no consensus regarding the best method to be used. Thus, the therapeutic choice should be individualized and defined according to the experience of the service, since each therapeutic option presents its advantages and disadvantages, and there is no medication considered superior and capable of eliminating HPV so far<sup>(7)</sup>.

Treatments may be topical (destructive by chemical, physical, immunomodulator or excisional agents) or systemic<sup>(14)</sup>. Conventional treatment consists of the lesion chemical destruction through topical application of podophylin, 5-fluorouracil, trichloroacetic acid and podophyllotoxin. Physical destruction methods include cryotherapy, laser, electrocauterization, and surgical excision<sup>(15)</sup>.

Topical treatments, such as imiquimod 5% cream and podophylin 20%, are options that should be considered, whenever possible, as they are less traumatic and have fewer sequelae. However, studies are conflicting and there is no consensus asserting that such methods are effective and free from adverse effects in patients under 12 years of age. At the time the patient was attended at our service, there were scarce studies that verified the safety of Imiquimod 5% in infants, and its use was disregarded at first<sup>(7,16,17)</sup>.

Another therapeutic method used is the high-frequency surgery, used mainly in those cases with numerous lesions, a situation whose topical treatment with chemical agents could be toxic due to the quantity required. This procedure was elected first in this study



Figure 4 – After the treatment with Imiquimode.

since this was the case of the patient reported. It consists of applying a high frequency current to excise the affected site. The technique is relatively simple and quite cost-effective, since it is performed with local anesthesia and without the need for a surgery center<sup>(18)</sup>.

As the treatment for high-frequency surgery did not show the expected effect, it was thought to use trichloroacetic acid, given the importance of its destructive therapeutic effect that causes the hydrolysis of cellular proteins, leading to apoptosis, without systemic toxicity (unlike other topical methods). Other effects consist mainly of local symptoms, such as pain at the site of application, burning, and hyperpigmentation<sup>(19,20)</sup>. It is important to stand out, however, that the patient, despite demonstrating discomfort during the application of the acid, did not evolve with alteration of perilesional skin staining.

In refractory cases to standard treatments, as the one of this study, the use of phytotherapy may be an alternative. *Thuja occidentalis* is a phytotherapeutic option described in the literature. In folk medicine, the extract has been used to treat repetition bronchitis, enuresis, cystitis, psoriasis, uterine carcinomas, amenorrhea, and rheumatologic symptoms. *T. occidentalis* can be used orally or by topical application. It is known to cause significant increase of some inflammatory mediators in order to activate a local inflammatory response, showing a possible antiviral activity. However, further studies are needed to clarify their exact action mechanism<sup>(21-24)</sup>.

Finally, due to the difficulty of complete resolution of the lesions based on this bibliography, the infant was submitted to imiquimode therapy. Imiquimod is a local application synthetic immunomodulating agent that acts on the innate and cellular immune system mediated by interferon alfa and tumor necrosis factor-alpha (IFN- $\alpha$ )<sup>(17)</sup>. Recent studies have shown that the imiquimod treatment in the child population shows good results, especially in lesions located in mucous membranes, where absorption is enhanced. The resolution of the lesions is usually obtained in 72 to 84% of the cases, and local recurrence rates range from 5 to 19%.

One of the main treatment advantages is the fact that it can be carried out at home and painless, as pain is a frequent symptom in destructive methods. Topical treatment is well tolerated, having as its main side effect contact dermatitis, occurring mainly in places of skin occlusion<sup>(17)</sup>. Although erythema is often reported in the literature, no important adverse effect occurred during the approach of the case presented in this report, and this was the definitive therapy for the resolution of recurrent lesions.

### CONCLUSION

The observation of this specific case suggests that imiquimod may be a safe and effective therapeutic alternative for the treatment of condyloma in the pediatric population. However, data regarding treatment in this age group are scarce, which hinders the conduct of refractory cases to those known to be safe and described by the literature. Thus, the importance of developing more studies on the theme is reinforced. Moreover, although vertical transmission is the main form of transmission, sexual abuse should always be considered, since only the diagnostic suspicion from the health professional, as happened in this reported experience, is able to facilitate the investigation and conduct of the case.

### Participation of each author

All authors contributed equally.

### Funding

The authors declare there was no financing of any kind for this research.

### **Conflict of interests**

The authors declare no conflict of interests.

### REFERENCES

- Marcoux D, Nadeau K, McCuaig C, Powell J, Oligny LL. Pediatric anogenital warts: a 7-year review of children referred to a tertiary-care hospital in Montreal, Canada. Pediatr Dermatol. 2006;23(3):199-207. https://doi.org/10.1111/j.1525-1470.2006.00218.x
- Rodrigues LRE, Portugal V, Rodrigues N, Nápoles S, Casanova C. Verrugas Anogenitais na Criança: A importância da Abordagem Multidisciplinar. Acta Med Port. 2011;24(2):367-70.
- Culton DA, Morrell DS, Burkhart CN. The management of condyloma acuminata in the pediatric population. Pediatr Ann. 2009;38(7):368-72. https://doi.org/10.3928/00904481-20090622-05
- Sinclair KA, Woods CR, Kirse DJ, Sinal SH. Anogenital and respiratory tract human papillomavirus infections among children: age, gender, and potential transmission through sexual abuse. Pediatrics. 2005;116(4):815-25. https://doi.org/10.1542/peds.2005-0652
- Unger ER, Fajman NN, Maloney EM, Onyekwuluje J, Swan DC, Howard L, et al. Anogenital human papillomavirus in sexually abused and nonabused children: a multicenter study. Pediatrics. 2011;128(3):e658-65. https://doi.org/10.1542/peds.2010-2247
- Costa-Silva M, Fernandes I, Rodrigues AG, Lisboa C. Anogenital warts in pediatric population. Ann Br Dermatol. 2017;92(5):675-81. http://dx.doi. org/10.1590/abd1806-4841.201756411
- Veasey JV, Dall'Antonia M, Miguel BAF, Mayor SAS, Campaner AB, Manzione TS. Condilomas anogenitais em crianças: análise descritiva de 20 casos. Surg Cosmet Dermatol. 2017;9(2):130-3. http://www.doi. org/10.5935/scd1984-8773.201792993
- Rizvi AA, Kanwar AJ, Goel S. Condyloma acuminata in a 3-year-old female: Sexual abuse or not? Indian J Pediatr Dermatol. 2016;17(3):221-2. http://doi.org/10.4103/2319-7250.179498
- Syrjänen S, Puranen M. Human papillomavirus infections in children: the potential role of maternal transmission. Crit Rev Oral Biol Med. 2000;11(2):259-74. https://doi.org/10.1177/10454411000110020801
- Thornsberry L, English III J. Evidence-based treatment and prevention of external genital warts in female pediatric and adolescent patients. J Pediatr Adolesc Gynecol. 2012;25(2):150-4. https://doi.org/10.1016/j. jpag.2011.10.004
- Percinoto ACC, Danelon M, Crivelini MM, Cunha RF, Percinoto C. Condyloma acuminata in the tongue and palate of a sexually abused child: a case report. BMC Res Notes. 2014;7:467. https://dx.doi. org/10.1186%2F1756-0500-7-467
- Hettler J, Greenes DS. Can the initial history predict whether a child with a head injury has been abused? Pediatrics. 2003;111(3):602-7. https://doi. org/10.1542/peds.111.3.602

- Swerdlin A, Berkowitz C, Craft N. Cutaneous signs of child abuse. J Am Acad Dermatol. 2007;57(3):371-92. https://doi.org/10.1016/j. jaad.2007.06.001
- Moresi JM, Herbert CR, Cohen BA. Treatment of anogenital warts in children with topical 0. 05% podofilox gel and 5% imiquimod cream. Pediatr Dermatol. 2001;18(5):448-50. https://doi.org/10.1046/j.1525-1470.2001.1980a.x
- Brás F, Sardinha R, Pacheco A. Modalidades terapêuticas no tratamento dos condilomas acuminados. Acta Obstet Ginecol Port. 2015;9(5):383-92.
- Forcier M, Musacchio N. An overview of human papillomavirus infection for the dermatologist: disease, diagnosis, management, and prevention. Dermatol Ther. 2010;23(5):458-76. https://doi.org/10.1111/j.1529-8019.2010.01350.x
- Brandt H, Patriota R, Belda Junior W, Fernandes J, Criado P. Tratamento do papiloma vírus humano na infância com creme de imiquimode a 5%. An Bras Dermatol. 2009;84(5):549-53. http://dx.doi.org/10.1590/S0365-05962010000400020
- Perlman SE, Lubianca JN, Kahn JA. Characteristics of a group of adolescents undergoing loop electrical excision procedure (LEEP). J Pediatr Adolesc Gynecol. 2003;16(1):15-20. https://doi.org/10.1016/ s1083-3188(02)00209-7
- Pezeshkpoor F, Banihashemi M, Yazdanpanah M, Yousefzadeh H, Sharghi M, Hoseinzadeh H. Comparative study of topical 80% trichloroacetic acid with 35% trichloroacetic acid in the treatment of the common wart. J Drugs Dermatol. 2012;11(11):e66-9.
- Jayaprasad S, Subramaniyan R, Devgan S. Comparative evaluation of topical 10% potassium hydroxide and 30% trichloroacetic acid in the treatment of plane warts. Indian J Dermatol. 2016;61(6):634-9. https:// doi.org/10.4103/0019-5154.193670
- Bodinet C, Lindequist U, Teuscher E, Freudenstein J. Effect of an orally applied herbal immunomodulator on cytokine induction and antibody response in normal and immunosuppressed mice. Phytomedicine. 2002;9(7):606-13. https://doi.org/10.1078/094471102321616418
- 22. Bodinet C, Freudenstein J. Effects of an orally applied aqueous-ethanolic extract of a mixture of Thujae occidentalis herba, Baptisiae tinctoriae radix, Echinaceae purpureae radix and Echinaceae pallidae radix on antibody response against sheep red blood cells in mice. Planta Med. 1999;65(8):695-9. https://doi.org/10.1055/s-1999-14044
- 23. Gohla SH, Haubeck H-D, Schrum S, Soltau H, Neth RD. Activation of CD4-positive T cells by polysaccharide fractions isolated from the Cupressaceae Thuja occidentalis L. (Arborvitae). In: Neth R, Gallo RC, Greaves MF, Gaedicke G, Gohla S, Mannweiler K, et al., editors. Modern Trends in Human Leukemia VIII. Berlin: Springer; 1989. p. 268-72.
- 24. Torres A, Vargas Y, Uribe D, Carrasco C, Torres C, Rocha R, et al. Proapoptotic and anti-angiogenic properties of the  $\alpha/\beta$ -thujone fraction from Thuja occidentalis on glioblastoma cells. J Neuro Oncol. 2016;128(1):9-19. https://doi.org/10.1007/s11060-016-2076-2

### Address for correspondence: GABRIEL DUQUE PANNAIN

Rua Itamar Soares de Oliveira, 211, apt. 202 – Cascatinha Juiz de Fora (MG), Brazil CEP: 36033-260 E-mail: gabrielduquep@gmail.com

Received on: 06.11.2019 Approved on: 09.30.2019

# GIANT CONDYLOMATA ACUMINATA OF BUSCHKE-LOWENSTEIN: Surgical method as effective treatment and Imiquimod as adjuvant therapy in an immunocompetent patient

Condiloma acuminado gigante de Buschke–Lowenstein: método cirúrgico como tratamento eficaz e imiquimode como terapia adjuvante em uma paciente imunocompetente

Helena Lucia Barroso dos Reis<sup>1</sup>, João Victor Jacomele Caldas<sup>2</sup>, Carolina Azevedo Feltz<sup>2</sup>, Danielle de Oliveira Machado<sup>2</sup>, Lucas Donateli Rosa, Marize de Freitas Santos Neves<sup>4</sup>, Dennis de Carvalho Ferreira<sup>5</sup>, Antônio Chambo Filho<sup>6</sup>, Neide Aparecida Tosato Boldrini<sup>4</sup>

### ABSTRACT

**Introduction:** Giant condylomata acuminata, also known as Buschke-Lowenstein tumor is a rare form of tumor of the anogenital condylomata acuminata, which is a sexually transmitted infection (STI) caused by the human papillomavirus (HPV). **Objective:** To report a case of giant condylomata acuminata in an immunocompetent patient. **Case report:** The patient was referred to the Outpatient Clinic for Sexually Transmitted Infections and AIDS at a public hospital in the city of Vitória, Espírito Santo State, Brazil, reporting the onset of progressive growth vertucous lesions on the external genitalia for four months. The patient underwent surgical ablation, and giant condylomata diagnostic confirmation was obtained through histopathology. She was treated with 5% imiquimod cream in routine applications for eight consecutive weeks to avoid recurrence and was also vaccinated for HPV after the procedure. **Conclusion:** Surgery excision is the treatment of choice in extensive genital condylomata lesions to exclude malignancy. Imiquimod use as adjuvant therapy for reducing recurrence seems to be adequate.

Keywords: Papillomavirus infections; imiquimod; condylomata acuminata.

### RESUMO

Introdução: O condiloma acuminado gigante, também conhecido tumor de Buschke-Lowenstein, é uma apresentação rara do condiloma acuminado anogenital, que é uma infecção sexualmente transmissível (IST) causada pelo papilomavírus humano (HPV). Objetivo: Relatar um caso de condiloma acuminado gigante em uma paciente imunocompetente. Relato de caso: A paciente foi encaminhada para o ambulatório de infecções sexualmente transmissíveis e AIDS de um hospital público na cidade de Vitória, Espírito Santo, Brasil, relatando o aparecimento de lesões verrucosas de crescimento progressivo na genitália externa por quatro meses. A paciente foi submetida à exérese cirúrgica e a confirmação diagnóstica de condiloma gigante foi obtida através da histopatologia. Ela foi medicada com imiquimode creme a 5% em aplicações rotineiras por oito semanas consecutivas para evitar recorrências e foi também vacinada contra o HPV após o procedimento. Conclusão: Exérese cirúrgica é o tratamento de escolha em lesões condilomatosas extensas para excluir malignidade. O uso de Imiquimode como terapia adjuvante para redução de recidivas mostrou-se adequado.

Palavras-chave: infecções por Papillomavirus; imiquimode; condiloma acuminado.

### INTRODUCTION

Giant condylomata acuminata, also known as Buschke-Lowenstein tumor, is a rare form of tumor of the anogenital condylomata acuminate<sup>(1)</sup>, which is a sexually transmitted infection (STI) caused by the human papillomavirus (HPV). Approximately 40 of known HPV have tropism for the anogenital site, and giant condylomata is associated with infection by HPV 6 and 11<sup>(2)</sup>. It is characterized by an extensive exophytic benign warty lesion

<sup>5</sup>Universidade Veiga de Almeida - Rio de Janeiro (RJ), Brazil.

6School of Sciences of the Santa Casa de Misericórdia de Vitória (EMES-

CAM), Santa Casa de Misericórdia de Vitória Hospital - Vitória (ES), Brazil.

with the potential for malignancy and local aggressive behavior<sup>(1)</sup>. Several therapeutic possibilities have been reported for the management of the giant condylomata acuminata, with variable outcomes. Surgical local excision is the treatment of choice in most hospital services, alone or in association with other treatment options like radiation therapy, topical and intralesional chemotherapy, systemic interferon  $\alpha$ -2b, carbon dioxide laser, and topical therapies including 5% imiquimod that can reduce recurrence, podophyllin and therapeutic vaccine, which is currently under research<sup>(2)</sup>. Although several warts are considered benign, a long-lasting genital lesion can turn malignant due to contact between the virus and the immunologic system. Thus, the time to remove the giant condylomata acuminata to prevent a carcinomatous transition needs to be prompt.

### **OBJECTIVE**

This study aimed to report a giant condylomata acuminata in an immunocompetent patient and to describe the therapeutic management.

<sup>&</sup>lt;sup>1</sup>Infectious Diseases Post-Graduate Program, Universidade Federal do Espírito Santo – Vitória (ES), Brazil.

<sup>&</sup>lt;sup>2</sup>Gynecology and Obstetrics Residency Program, Universidade Federal do Espírito Santo – Vitória (ES), Brazil.

<sup>&</sup>lt;sup>3</sup>Gynecology and Obstetrics Residency Program, UNIMED Vitória – Vitória (ES), Brazil.

<sup>&</sup>lt;sup>4</sup>Gynecology and Obstetrics Department, Universidade Federal do Espírito Santo – Vitória (ES), Brazil.

### CASE REPORT

A 21-year-old female patient, married for about six months, farmer, with incomplete high school, and no condom use in oral and vaginal sexual intercourse was referred to the Outpatient Clinic for Sexually Transmitted Infections and AIDS at a public hospital in the city of Vitória, Espírito Santo State, Brazil, reporting onset of progressive growth verrucous lesions on the external genitalia for four months. She was treated with benzathine penicillin and oral metronidazole previously and denied smoking, drinking, or drug use. The tumor was painful and bleeding due to secondary bacterial infection. Physical examination revealed bilateral coalescent verrucous lesion occupying the whole extension of labia majora and minora as well as perianal lesions clinically compatible with Buschke-Lowenstein giant genital condylomata (**Figure 1**).

Cervical examination was unchanged, and cervical and vaginal cytology, colposcopy, and otolaryngologic examination were unchanged; moreover, serological tests for HIV, viral hepatitis, and syphilis presented no abnormalities. The sexual partner underwent serological tests that were unchanged, and genital inspection did not present condylomata lesions. Hospitalization was indicated, and gentamicin was administered for two weeks. The patient underwent a surgical ablation, and giant condylomata diagnostic confirmation was obtained through histopathology, which showed hyperkeratosis and papillomatosis with multiple koilocytes, with fast mitotic activity and no cellular atypia, suggestive of HPV infection (**Figure 2**).

HPV 11 was identified by the PCR test. After surgery, in a 30-day follow-up consultation, she presented persistent perianal lesions (**Figure 3**). The patient was submitted to 70% trichloroacetic acid application in the perianal lesions for seven weeks, with resolution of the condition. She underwent the use of 5% imiquimod cream in routine applications for eight consecutive weeks to avoid recurrence (**Figure 4**) and was vaccinated for HPV after the procedure. On her three-year follow-up, she remained disease-free, having periodic consultations, using a barrier method to prevent new STI and without any recurrence.

The study project was submitted to the Ethical Committee for Research of the Federal University of Espírito Santo.



Figure 1 – Giant condylomata acuminata of the vulva. Large, irregular, ulcerative cauliflower-like tumor, covering the vulvar and perianal area.



**Figure 2** – Histopathological findings showing hyperkeratosis and papillomatosis with multiple koilocytes, consistent with HPV infection (hematoxylin and eosin staining, ×10).



Figure 3 - Persistent perianal lesion after vulvar electrosurgical resection.



**Figure 4** – Patient's vulva, six months after surgery. Complete resolution of the genital condylomata.

### DISCUSSION

Giant condylomata acuminata is a very rare clinical condition<sup>(1)</sup>. Currently, surgery can be avoided in some patients due to developed topical therapeutic methods<sup>(3)</sup>. Conversely, surgical excision remains the standard therapy in cases of high-risk condylomata acuminata or if topical therapy appears unworkable due to the size of the affected lesion<sup>(3,4)</sup>. Here, a case of a rapid-growth HPV lesion in a female patient was reported, who was very embarrassed about her condition and lived in a difficult access region. The condition, HPV 11, was identified by PCR test, in concordance with literature reviews<sup>(2,3)</sup>. The patient was evaluated for STI and cervical cancer prevention due to its HPV association, as cytological screening to detect precursor lesions is still paramount in some Brazilian regions<sup>(5)</sup>. She underwent high-frequency surgical excision, which is indicated for extensive condylomata lesions due to its advantages in controlling hemostasis, resolution rates in 90% of cases, and recurrence of about 25%<sup>(2-4)</sup>. Moreover, surgical ablation is the first option to exclude malignancy and to avoid the development of giant condilomata<sup>(4,6)</sup>. Recurrences may occur in 25 to 66% of cases after surgical treatment, according to some authors<sup>(3,7)</sup>, which are avoided by use of imiquimod after surgery in regular applications for eight consecutive weeks as well as by HPV vaccination. HPV vaccines have been known for their protecting results on cervical cancer and genital warts, though the treatment outcome is still indefinite. Therapeutic vaccine for HPV aims to generate cell-mediated immunity and studies on the subject are in progress<sup>(8,9)</sup>. Currently, a prospective study suggested that HPV vaccines could be effective in the management of genital warts<sup>(9)</sup>.

Imiquimod, an immune response modulator, has also been used, in regular applications, as an alternative therapy in HPV lesions by inducing the production of interferon alfa and cytokines, enhancing the immune response against HPV infected cells<sup>(3,10)</sup>.

Therefore, the association of surgical techniques and topical 5% imiquimod therapy was successful in this case report, highlighting that the controlling needs to be individualized.

### CONCLUSION

Surgery excision is the treatment of choice in extensive genital condylomata lesions and should be readily instituted to exclude malignancy. Imiquimod use as adjuvant therapy for reduced recurrence seems to be appropriate, as in the present case.

### Participation of each author

Helena Lucia Barroso dos Reis, Antônio Chambo Filho, and Neide Aparecida Tosato Boldrini conceived the study, collected data, and analyzed and interpreted clinical data. Antônio Chambo Filho, Helena Lucia Barroso dos Reis, and Neide Aparecida Tosato Boldrini attended to the patient. João Victor Jacomele Caldas, Carolina Azevedo Feltz, Danielle de Oliveira Machado, and Lucas Donateli Rosa performed the systematic review. Dennis de Carvalho Ferreira and Marize de Freitas Santos Neves performed laboratorial studies. Helena Lucia Barroso dos Reis and Antônio Chambo Filho developed the first draft. All authors approved the subsequent draft versions. All authors approved the final submitted version.

### Funding

No specific funding was received for this study.

### **Conflict of interests**

The authors declare no conflict of interests.

### **Ethical approval**

The study was approved by the Ethics Committee of the Universidade Federal do Espírito Santo and informed consent was waived for this study.

### REFERENCES

- Sandoval I, Hernández R, Torres E, Yanque O. Giant condylomata acuminata of Buschke-Lowenstein. J Obstet Gynaecol. 2019:1-2. https:// doi.org/10.1080/01443615.2019.1607834
- Nadal SR, Manzione CR, Horta SHC, Calore EE. Sistematização do atendimento dos portadores de infecção perianal pelo papilomavirus humano (HPV). Rev Bras Coloproct. 2004;24(4):322-8.
- Spinu D, Rădulescu A, Bratu O, Checheriță IA, Ranetti AE, Mischianu D. Giant condyloma acuminatum - Buschke-Lowenstein disease – a literature review. Chirurgia (Bucur). 2014;109(4):445-50.
- Koukoura O, Klados G, Strataki M, Daponte A. A rapidly growing vulvar condylomaacuminatum in a young patient. BMJ Case Rep. 2015;2015. https://doi.org/10.1136/bcr-2014-208126
- Boldrini NT, Freitas LB, Coutinho AR, Loureiro FZ, Spano LC, Miranda AE. High-grade cervical lesions among women attending a reference clinic in Brazil: associated factors and comparison among screening methods. PLoS One. 2014;9(7):e102169. https://doi.org/10.1371/journal. pone.0102169
- Chu QD, Vezeridis MP, Libbey NP, Wanebo HJ. Giant condyloma acuminatum (Buschke-Lowenstein tumor) of the anorectal and perianal regions: analysis of 42 cases. Dis Colon Rectum. 1994;37(9):950-7. https://doi.org/10.1007/bf02052606
- Papiu HS, Duminici A, Olariu T, Onita M, Hornung E, Goldis D, et al. Perianal giant condylomaacuminatum (Buschke-Löwenstein tumor). Case report and review of literature. Chirurgia (Bucur). 2011;106(4):535-9.
- Yang A, Farmer E, Wu TC, Hung CF. Perspectives for therapeutic HPV vaccine development. J Biomed Sci. 2016;23(1):75. https://doi. org/10.1186/s12929-016-0293-9
- Choi H. Can quadrivalent human papillomavirus prophylactic vaccine be an effective alternative for the therapeutic management of genital warts? An exploratory study. Int Braz J Urol. 2019;45(2):361-8. https://doi. org/10.1590/S1677-5538.IBJU.2018.0355
- Passos MRL, editor. DST, Doenças Sexualmente Transmissíveis. 5<sup>a</sup> ed. Rio de Janeiro: Cultura Médica; 2006.

### Address for correspondence: HELENA LUCIA BARROSO DOS REIS

Ambulatório de Ginecologia e Obstetrícia, Universidade Federal do Espírito Santo Avenida Marechal Campos, 1468 – Maruípe

Vitória (ES), Brazil CEP: 29043-260

E-mail: dr.hbarroso@gmail.com

Received on: 06.07.2019 Approved on: 07.19.2019

# NONAVALENT HPV VACCINE: WHAT ARE THE ADVANTAGES?

VACINA NONAVALENTE CONTRA O HPV: QUAIS SÃO AS VANTAGENS?

Marília Barcelos Marqui<sup>1</sup>, Luana Tiuma Borba<sup>1</sup>, Luiza de Souza Duarte<sup>1</sup>, Vinicius D'Ávila Bitencourt Pascoal<sup>1,2</sup>, Fabiana Nunes Germano<sup>1,3</sup>, Aislan Cristina Rheder Fagundes Pascoal<sup>1,2</sup>

### **INTRODUCTION**

Sexually transmitted infections caused by the human papillomavirus (HPV) are the most prevalent in the world and can cause from genital warts to premalignant lesions, which can lead to cervical cancer, vaginal and vulvar neoplasms, anal and penile cancer, if left untreated. Non-genital pathologies may also be related to HPV, such as oropharyngeal cancer and recurrent respiratory papillomatosis<sup>(1-3)</sup>.

HPV can be transmitted whenever there is direct contact of infected hair and mucosa during sexual intercourse. It can also occur via the maternal fetal route (pregnancy, intra and peripartum)<sup>(4)</sup>. The lesions are imperceptible in many cases, without manifestations of infection, which may facilitate transmission.

This virus affects men and women and is found not only in anogenital regions, but also in extragenital ones. Infection can be manifested in clinical, subclinical and latent forms<sup>(5)</sup>, with subclinical and asymptomatic forms predominating in men<sup>(6)</sup>.

Around 90% of the HPV infections are eliminated in almost two years and, therefore, there is no need of any pharmacological treatment<sup>(7)</sup>. However, when there is a persistent infection, different pathologies may be present depending on the virus sub-genotype. Infections caused by low-risk oncogenic sub-genotypes may be related to genital warts and respiratory papillomatosis, and highrisk oncogenic subtypes may be associated with the pathogenesis of cervical, anal, penile, vulva, and vaginal and oropharyngeal cancer. Studies have also associated HPV with other cancers, such as oral cavity squamous cell carcinoma<sup>(8)</sup>.

In an attempt to reduce the number of HPV-associated pathologies, there are three vaccines present in the world market currently: Cervarix<sup>®</sup> (bivalent vaccine) produced by GlaxoSmithKline; Gardasil<sup>®</sup> (quadrivalent vaccine) by Merck, and most recently, the Gardasil 9<sup>®</sup> (nonavalent vaccine) was released. All three vaccines are highly immunogenic, effective and safe. The only difference between them is the HPV sub-genotypes that are present<sup>(9)</sup>. HPV vaccination has been known as a mean to significantly reduce the incidence of related anogenital cancers and the appearance of genital warts<sup>(10)</sup>.

In Brazil, the quadrivalent vaccine (Gardasil<sup>®</sup>) is available free of charge in the Unified Health System (SUS) units for girls aged 9 to

14 and boys aged 11 to 14. In 2018, the second-generation Gardasil, Gardasil9<sup>®</sup>, was approved by ANVISA (Brazilian Health Surveillance Agency). Considering the world scenario, in which HPV has been associated with pathologies of a large number of morbidities and mortality, does the nonavalent vaccine have advantages over the other vaccines in the market?

This paper aims to evaluate the advantages and/or disadvantages of the nonavalent vaccine in relation to the other two vaccines by performing a literature review, using Scielo and PubMed databases and the keywords: "HPV", "Vaccine and HPV", "Nonavalent Vaccine and HPV", "Quadrivalent Vaccine and HPV", in Portuguese and English.

### HUMAN PAPILLOMAVIRUS

HPV are viruses belonging to the Papillomaviridae family, which has over 39 genera. The classification by type is based on the comparison of L1 gene nucleotide sequences of the various HPVs, each type differs from the other types by at least 10% in the sequence of L1 nucleotides.<sup>(11,12)</sup>. Papillomaviruses that share an identity of 60 to 70% in this gene are designated as species, while types in a given species have show an identity between 71 and 89%<sup>(1,13)</sup>. The genera Alpha, Beta, Um and Nupapillomavirus are capable of infecting humans and have more than 200 types<sup>(12-14)</sup>. They are viruses of about 55 nm long, with a double-stranded DNA genome of approximately 5 to 8 Kb that is not coated with a lipoprotein envelope. The viral genome may be present in the episomal (circular) form associated with nuclear histones in benign lesions, or it may be integrated into the host cell genome as in malignant lesions. In the latter case, there is a greater risk of cell transformation as a result of early viral protein expression<sup>(15)</sup>.

The genome has about ten open reading frames (ORF) that consist of an early region (E) formed by proteins expressed in the early phase of the replicative cycle, responsible for viral transcription and replication. The late region (L), responsible for the proteins expressed in the later phase of the HPV cycle, are the structural proteins. There is also a long control region (CSF), which is important for the regulation of viral genome transcription and replication events and where the origin of replication is located. The early proteins are E1, E2, E4, E5, E6, and E7. These proteins can interfere in the host cell to the point of immortalizing it. The E6 and E7 oncoproteins direct the degradation of tumor suppressor proteins p53 and pRB, respectively. Therefore, they can contribute to the progression of the carcinogenesis process and genomic instability<sup>(16,17)</sup>. Late proteins are L1 and L2. They are structural proteins that form the capsid and are responsible for the virus immunogenicity and for carrying gender-specific antigenic determinants(11,12,14).

<sup>&</sup>lt;sup>1</sup>Grupo HPV sem Neura, Laboratório Multiusuário de Pesquisa Biomédica, Departamento de Ciências Básicas, Instituto de Saúde de Nova Friburgo, Universidade Federal Fluminense – Nova Friburgo (RJ), Brazil. <sup>2</sup>Programa de Pós-graduação em Ciências e Biotecnologia, Instituto de Biologia, Universidade Federal Fluminense – Niterói (RJ), Brazil. <sup>3</sup>Programa de Pós-graduação em Odontologia, Instituto de Saúde de Nova Friburgo, Universidade Federal Fluminense (UFF) – Nova Friburgo (RJ), Brazil.

HPV has tropism for cutaneous, mucosal, and cervical epithelial cells, preferentially infecting the most basal layers, in which there is greater cell proliferation<sup>(15)</sup>. Based on the ability to produce malignant lesions, HPV can be divided into low, moderate and high risk types. Persistent HPV-16 infection is the most potent type among the neoplasm-associated viral types. It is known to lead to the development of cancer in various body regions, resulting in infected cells to immortalize genomic instability, inhibit response to DNA damage and escape from apoptosis<sup>(18)</sup>. Types 18, 31, 33, 35, 39, 45, 51, 52, 56, 58, and 59 are also associated with cancer, especially cervical cancer.

Type 68 has limited evidence in human studies, but strong mechanistic evidence of cervical cancer. Types 26, 53, 66, 67, 70, 73 and 82 present limited evidence in humans in the development of cervical cancer. Finally, types 6 and 11, which belong to the alpha papillomavirus ten species, are classified as non-carcinogenic to humans based on epidemiological evidence and absence of carcinogenic potential in the studied mechanisms<sup>(18)</sup>.

### Epidemiology

Around 15% of all human cancers are believed to be caused by viral infections, and 5% may be attributed to HPV infection. Regarding cervical cancer, about 99% of cases are related to HPV<sup>(1-3,7-9)</sup>. This persistent infection, in turn, can lead to the development of several neoplastic forms, in addition to cervical cancer, such as vagina, vulva, penile, anus, and oropharynx<sup>(1-3,7-9)</sup>. Cervical cancer is among the most prevalent cancers in the world, with 530,000 new cases per year<sup>(19)</sup>.

In Brazil, cervical cancer is the third most common tumor in the female population and the fourth leading cause of death in women, with 16,340 (7.9%) new cancer cases of cervical cancer and 5,430 deaths of women per year, according to the National Cancer Institute<sup>(20)</sup>.

Around half of women diagnosed with cervical cancer are under 50 years old, and more than two thirds are diagnosed in less developed countries. Southeast Asia (especially India), Latin America, and sub-Saharan Africa are areas with the highest percentages of cases. HPV 16 and 18 together account for 71% of cervical cancer globally. This percentage increases to 90% when we include HPV types 6, 11, 16, 18, 31, 33, 45, 52 and 58<sup>(19)</sup>.

Recent studies have been conducted to evaluate the prevalence of the young Brazilian population. The study population conducted by the Brazilian Department of Health consisted of 5,812 women and 1,774 men, with a mean age of 20.6. The city of Salvador had 71.9% of HPV-positive participants, and the lowest rate was in the city of Recife, with 41.2% of survey participants<sup>(21)</sup>.

The distribution and prevalence of HPV types vary based on the degree of cervical disease, age, and geographic location of patients<sup>(22-25)</sup>. The prevalence of HPV 16 and 18 has been observed in more severe lesions, but they are also the most frequent types observed in women with normal cytology<sup>(26)</sup>. A recent study carried out in Ghana showed that infection with high-risk types alone was seen in 32.3% of participants, while infection with multiple highrisk types was 9.7%.

The five most common HPV types detected were: 16 (7.4%), 52 (7.2%), 35 (4.8%), 59 (4.7%), 56  $(3.9\%)^{(27)}$ . HPV-18 has been found in many regions as the second most common and in some

Brazilian studies as well<sup>(28-31)</sup>. HPV-31 ranks second in frequency in Europe, HPV-52 is especially frequent in Africa and North America, and HPV-45 is in third place in Africa, Central/West Asia, North America, and Oceania<sup>(28,32)</sup>. In Brazil, HPV 31 and 33 are the second most prevalent types in the Northeast and Central regions<sup>(33,34)</sup>.

In Mato Grosso do Sul, HPV-66 was detected in 22% of HPVpositive samples<sup>(35)</sup>, and HPV-58 was the most frequent type (19.8%) in HIV-infected women followed by HPV-53 (15.5%) in Rio de Janeiro<sup>(36)</sup>. HPV DNA was detected in 67.5% of women. In a study in the state of São Paulo, HPV-16 (40%) was the most prevalent type in most patients with lesions, followed by HPV 31 (13.3%), 45 (13.3%) and 18 (4.1%). Multiple infections occurred in 15% of the cases, and infections with other types of HPV were detected in 14% of the samples. Thus, HPV 16 and 18 infections do not always occur in a solitary manner (single infection) and are associated with other types of HPV on several occasions<sup>(37)</sup>.

### **HPV VACCINES**

Three types of the HPV vaccine are available for the prevention of HPV-related diseases nowadays. The bivalent vaccine (Cervarix<sup>®</sup>) acts against HPV types 16 and 18. The quadrivalent vaccine targets types 6, 11, 16 and 18 (Gardasil<sup>®</sup>). More recently, the nonavalent vaccine (Gardasil 9<sup>®</sup>), which includes the most oncogenic genotypes prevalent and relevant low risk types 6, 11, 16, 18, 31, 33, 45, 52 and 58<sup>(38)</sup>, was released.

### **Bivalent vaccine**

This vaccine contains the viral capsid protein L1 and is produced by recombinant DNA technology using the *Trichoplusnia ni* insect cell baculovirus expression system to obtain VLPs, which are analogous viral particles of the two most common types present in neoplasms, cervical lesions, HPV-16 and HPV-18, accounting for 70% of the cases of this type of cancer<sup>(38)</sup>.

Stimulation to the organism by the viral subtypes present in the VLP triggers the production of neutralizing antibodies against them, which provides the protection by the vaccine<sup>(39)</sup>. The adjuvant used in this vaccine is ASO4 with 500  $\mu$ g aluminum hydroxide and 50  $\mu$ g 3-deacylated monophosphoryl lipid-A<sup>(38)</sup>. Cervarix<sup>®</sup> is not a replicative vaccine and may be given together with other inactivated or replicative vaccines. However, its use is not recommended in pregnant women<sup>(38)</sup>.

Side effects include injection site reactions, such as pain (91.8%), redness (48%), and edema (44.1%). Some people who received the vaccine described fatigue (55%), myalgia (49%), arthralgia (21%), and hives (7%) after immunization. In approximately 6,400 patients who received the vaccine, 28% reported gastrointestinal symptoms of nausea, vomiting, diarrhea, and abdominal pain, and 13% developed fever within seven days after vaccination<sup>(40)</sup>.

### Quadrivalent vaccine

The quadrivalent HPV vaccine (Gardasil-4<sup>®</sup>) is also produced by a recombinant DNA technique in *Saccharomyces cerevisiae* yeast using the virus capsid protein L1, resulting in the production of virus-like particles (VLP). These VLP resemble the virus, with the same shape and size, but are not infectious because they do not have genetic material. The quadrivalent vaccine is composed of HPV 6, 11, 16, and 18 capsid L1 proteins<sup>(39,41)</sup>. Protection is conferred by stimulating the body's production of neutralizing antibodies against viral subtypes present in the VLP<sup>(39)</sup>.

It is known that the vaccine protects the HPV genotypes present in its cross-protection between other viral types due to gene similarity. The quadrivalent vaccine appears to provide partial cross-protection (around 59%) against types 31 and 45. Although these data have not been confirmed yet, there is also evidence of cross-protection against types 33, 52, and  $58^{(41)}$ . Studies have also shown a 38% efficacy of this vaccine against subtypes 31, 33, 35, 39, 45, 51, 52, 56, 58 and 59 in preventing cervical intraepithelial neoplasms 2 and 3 and adenocarcinoma *in situ*<sup>(42)</sup>.

The quadrivalent vaccine, or Gardasil<sup>®</sup>, was approved by Anvisa in 2006. Brazil was one of the first countries to approve the HPV vaccine. It was implemented in the national immunization program by the Brazilian Department of Health in SUS in 2014. When it was released, the vaccine was recommended only for girls aged 11 to 13, but it has been currently applied to girls aged between 9 and 14 and boys from 11 to 14 years old<sup>(41)</sup>.

### Nonavalent vaccine

Following the first licensed Gardasil-4<sup>®</sup> (first generation vaccine), the second generation vaccine was Gardasil-9<sup>®</sup>, which covers types 6, 11, 16, 18, 31, 33, 45, 52, and 58, and these types together account for 90% of cervical cancer cases and other intraepithelial neoplasms. This increases the protection of the population against persistent infections that could lead to some mucous or skin cancers<sup>(43)</sup>.

It is also produced by a recombinant DNA technique through expression in *Saccharomyces cerevisiae* yeasts, and VLP formed by capsid proteins of nine distinct viral subtypes will protect against the HPV types associated with approximately 90% of cervix cancer cases in women and 80 to 95% of other HPV-associated anogenital cancers in men and women<sup>(43,45)</sup>.

The nonavalent vaccine, or Gardasil-9<sup>®</sup>, was approved by the US Food and Drug Administration (FDA) in 2014. In 2017, the second-generation vaccine was approved and registered by Anvisa and was indicated for girls and women, as well as for boys and men aged 9 to 26<sup>(46)</sup>.

In a study by the Merck Sharp & Dohme Pharmaceuticals, comparing the two vaccines and highlighting the advantages of the second-generation vaccine, the Gardasil-9<sup>®</sup> not only presents the same benefits as Gardasil-4<sup>®</sup>, it also promotes immunization of the population of five types. This is more than the first-generation vaccine, showing better results against HPV 31, 33, 45, 52 and 58 related to undetermined atypical squamous cells and positive high-risk HPV or worse abnormality in the Pap smear<sup>(47)</sup>. A major advantage is also the extended age range indicated for vaccination, as the age range for Gardasil-4<sup>®</sup> vaccination is 9 to 26 years and is indicated for women and men older than 45<sup>(48)</sup>.

In Brazil, surveys were conducted to estimate the prevalence of new cases per 100,000 inhabitants, with the highest prevalence recorded in states with the lowest level of regional development. This is a national profile that follows the world profile, where the highest prevalence rates are found in the in the Midwest, North, Northeast, Southeast and South regions, respectively. The most common oncogenic viral types in Brazil are HPV 6, 11, 16, 18, and 31<sup>(49)</sup>.

### COST-EFFECTIVENESS

There are 13 types of HPV known to cause cervical cancer and contribute to cancer in the anogenital region, vagina, vulva, anus, and penis, as well as in the oropharynx, mainly tonsillar region and tongue base<sup>(19,50)</sup>. Estimates of the possible health and economic effects of HPV vaccination provide vital evidence to support the introduction of the vaccine into national programs<sup>(51)</sup>. To assess the cost and benefit ratio of HPV vaccination, a quality-adjusted incremental cost per year of life (QALY) calculation is made from the expanded scenario, assessing the vaccination costs and those related to associated pathologies<sup>(52)</sup>. Recent work has shown a gain in QALY, and when vaccination occurs in both men and women, this gain is even greater<sup>(51-53)</sup>.

Thus, vaccination, even with all related expenses, leads to a gain in public health. The nonavalent vaccine is an important advance, as the nine types present in the vaccine account for about 90% of cervical cancer cases. Non-vaccination vaccination can bring substantial additional public health benefits and is a very positive cost-effective intervention.

### CONCLUSION

The great advantage of the nonavalent vaccine is protection of types that are represented among the most prevalent in some Brazilian regions, which were not present in the quadrivalent vaccine. Thus, it is believed that the nonavalent vaccine may present a new tool to control cases of persistent infections by high-risk types not covered by the quadrivalent vaccine and, in the long term, lead to a decrease in the number of cases of cervical cancer and other HPV-related cancers. We hope this paper will be important to disseminate knowledge of the new HPV vaccine efficacy. Further studies are expected to be conducted so this vaccine may be soon implemented in the National Immunization Program, replacing the quadrivalent vaccine, protecting the population against the most common infections and preventing injuries caused by HPV.

### Participation of each author

Aislan Cristina Rheder Fagundes Pascoal was responsible for project design, literature review, manuscript writing. Marília Barcelos Marqui, Luana Tiuma Borba and Luiza de Souza Duarte were responsible for literature review and manuscript writing. Vinicius D'Ávila Bitencourt Pascoal and Fabiana Nunes Germano were responsible for manuscript revision.

### Funding

None.

### **Conflict of interests**

The authors declare no conflict of interests.

### **Ethics Committee Approval for Human Research**

It is not applicable because it is a bibliographic review.

### REFERENCES

- Bernard HU, Burk RD, Chen Z, van Doorslaer K, zur Hausen H, de Villiers EM. Classification of papillomaviruses (PVs) based on 189 PV types and proposal of taxonomic amendments. Virology. 2010;401(1):70-9. https://dx.doi.org/10.1016% 2Fj.virol.2010.02.002
- Chelimo C, Wouldes TA, Cameron LD, Elwood JM. Risk factors for and prevention of human papillomaviruses (HPV), genital warts, and cervical cancer. J Infect. 2013;66(3):207-17. https://doi.org/10.1016/j. jinf.2012.10.024
- Schiffman M, Doorbar J, Wentzensen N, de Sanjosé S, Fakhry C, Monk BJ, et al. Carcinogenic human papillomavirus infection. Nat Rev Dis Primers. 2016;2:16086. https://doi.org/10.1038/nrdp.2016.86
- Queiroz AMA, Cano MAT, Zaia JE. O papiloma vírus humano (HPV) em mulheres atendidas pelo SUS, na cidade de Patos de Minas - MG. Rev Bras Anál Clín. 2007;39(2):151-7.
- Carvalho ALS, et al. Sentimentos vivenciados por mulheres submetidas a tratamento para papillomavirus humano. Esc Anna Nery. 2007;11(2):248-53. http://dx.doi.org/10.1590/S1414-81452007000200010
- Costa LA, Goldenberg P. Papilomavírus humano (HPV) entre jovens: um sinal de alerta. Saúde Soc. 2013;22(1):249-61. https://dx.doi.org/10.1590/ S0104-12902013000100022
- Ho GY, Bierman R, Beardsley L, Chang CJ, Burk RD. Natural history of cervicovaginal papillomavirus infection in young women. N Engl J Med. 1998;338(7):423-8. https://doi.org/10.1056/NEJM199802123380703
- Chang F, Syrjänen S, Kellokoski J, Syrjänen K. Human papillomavirus (HPV) infections and their associations with oral disease. J Oral Pathol Med. 1991;20(7):305-17. https://doi.org/10.1111/j.1600-0714.1991. tb00936.x
- Oberlin AM, Rahangdale L, Chinula L, Fuseini NM, Chibwesha CJ. Making HPV vaccination available to girls everywhere. Int J Gynecol Obstet. 2018;143(3):267-76. https://doi.org/10.1002/ijgo.12656
- Immunization Expert Work Group, Committee on Adolescent Health Care. Committee Opinion No. 704: Human Papillomavirus Vaccination. 2017;129(6):e173-e178. https://doi.org/10.1097/AOG.00000000002052
- Camara GNN de L, Cruz MR, Veras VS, Martins CRF. Os papilomavírus humanos – HPV: histórico, morfologia e ciclo biológico. Universitas Ciênc Saúde. 2003;1(1):149-58. http://doi.org/10.5102/UCS.V111.502
- Bernard HU. Taxonomy and phylogeny of papillomaviruses: An overview and recent developments. Infect Genet Evol [Internet]. 2013 [cited on Aug. 8, 2019];18:357-61. Available at: http://dx.doi.org/10.1016/j. meegid.2013.03.011
- Harden ME, Munger K. Human Papillomavirus Molecular Biology. Mutation Res [Internet]. 2017 [cited on Aug. 8, 2019];772(3-12). Available at: http://dx.doi.org/10.1016/j.mrrev.2016.07.002
- International Committee on Taxonomy of Viruses. Taxonomy [Internet]. [cited on Jul. 31, 2019]. Available at: https://talk.ictvonline.org/taxonomy/
- Rosa MI, Medeiros LR, Rosa DD, Bozzeti MC, Silva FR, Silva BR. Human papillomavirus and cervical neoplasia. Cad Saúde Pública. 2009;25(5). http://doi.org/10.1590/S0102-311X2009000500002
- Vande Pol SB, Klingelhutz AJ. Papillomavirus E6 oncoproteins. Virology. 2013;445(1-2):115-37. https://doi.org/10.1016/j.virol.2013.04.026
- Roman A, Munger K. The papillomavirus E7 proteins. Virology. 2013;445(0):138-68. https://dx.doi.org/10.1016%2Fj.virol.2013.04.013
- Bouvard V, Baan R, Straif K, Grosse Y, Secretan B, Ghissassi F, et al. A review of human carcinogens — Part B: biological agents. Lancet Oncol. 2009;10(4):321-2. https://doi.org/10.1016/s1470-2045(09)70096-8
- Martel C, Plummer M, Vignat J, Franceschi S. Worldwide burden of cancer attributable to HPV by site, country and HPV type. Int J Cancer. 2017;141(4):664-70. https://doi.org/10.1002/ijc.30716
- INCA. Câncer de colo de útero [Internet]. INCA; 2016 [cited on Jun. 13, 2019]. Available at: https://www.inca.gov.br/tipos-de-cancer/cancer-docolo-do-utero

- Brasil. Ministério da Saúde. Estudo apresenta dados nacionais de Prevalência da Infecção pelo HPV [Internet]. Ministério da Saúde; 2017 [cited on Jun. 12, 2019]. Available at: http://www.saude.gov.br/noticias/ agencia-saude/42003-estudo-apresenta-dados-nacionais-de-prevalenciada-infeccao-pelo-hpv
- Fernandes JV, Meissner R de V, de Carvalho MG, Fernandes TA, de Azevedo PR, Villa LL. Prevalence of HPV infection by cervical cytologic status in Brazil. Int J Gynaecol Obstet. 2009;105(1):21-4. https://doi. org/10.1016/j.ijgo.2008.12.004
- Otero-Motta AP, Ordóñez JL, González-Celador R, Rivas B, Macías M del C, Bullón A, et al. Prevalence of human papillomavirus genotypes in cytologic abnormalities from unvaccinated women living in north-western Spain. APMIS. 2011;119(3):204-15. http://doi.org/10.1111/j.1600-0463.2010.02711.x
- Levi JE, Longatto-Filho A, Eluf-Neto J, Rodrigues CL, Oliveira CM, Carloni AC, et al. Evaluation of HPV molecular tests in primary screening for cervical cancer in Brazil. Open J Obstet Gynecol. 2014;4(8):470-8. http://doi.org/10.4236/ojog.2014.48068
- Bruno A, Serravalle K, Travassos AG, Lima BGC. Distribuição dos Genótipos de papillomavírus humano em mulheres do estado da Bahia, Brasil. Rev Bras Ginecol Obstet. 2014;36(9):416-22. http://doi. org/10.1590/SO100-720320140004995
- de Sanjosé S, Diaz M, Castellsagué X, Clifford G, Bruni L, Muñoz N, et al. Worldwide prevalence and genotype distribution of cervical human papillomavirus DNA in women with normal cytology: a meta-analysis. Lancet Infect Dis. 2007;7(7):453-9. http://doi.org/10.1016/S1473-3099(07)70158-5
- Krings A, Dunyo P, Pesic A, Tetteh S, Hansen B, Gedzah I, et al. Characterization of Human Papillomavirus prevalence and risk factors to guide cervical cancer screening in the North Tongu District, Ghana. PLoS One. 2019;14(6):e0218762. http://doi.org/10.1371/journal.pone.0218762
- Bruni L, Diaz M, Castellsagué X, Ferrer E, Bosch FX, de Sanjosé S. Cervical human papillomavirus prevalence in 5 continents: meta-analysis of 1 million women with normal cytological findings. J Infect Dis. 2010;202(12):1789-99. http://doi.org/10.1086/657321
- Noronha V, Mello W, Villa L, Brito A, Macêdo R, Bisi F, et al. Human papillomavirus associated with uterine cervix lesions. Rev Soc Bras Med Trop. 1999;32(3):235-40. http://doi.org/10.1590/S0037-86821999000300003
- Eluf-Neto J, Booth M, Muñoz N, Bosch FX, Meijer CJ, Walboomers JM. Human papillomavirus and invasive cervical cancer in Brazil. Br J Cancer. 1994;69:114-9. http://doi.org/10.1038/bjc.1994.18
- Oliveira-Silva M, Lordello CX, Zardo LM, Bonvicino CR, Moreira MA. Human Papillomavirus in Brazilian women with and without cervical lesions. Virol J. 2011;8:4. http://doi.org/10.1186/1743-422X-8-4
- Bruni L, Diaz M, Castellsagué X, Ferrer E, Bosch FX, de Sanjosé S. Cervical human papillomavirus prevalence in 5 continents: meta-analysis of 1 million women with normal cytological findings. J Infect Dis. 2010;202(12):1789-99. http://doi.org/10.1086/657321
- Baldez da Silva MF, Chagas BS, Guimarães V, Katz LM, Felix PM, Miranda PM, et al. HPV31 and HPV33 incidence in cervical samples from women in Recife, Brazil. Genet Mol Res. 2009;8(4):1437-43. http://doi. org/10.4238/vol8-4gmr677
- Rabelo-Santos SH, Zeferino L, Villa LL, Sobrinho JP, Amaral RG, Magalhães AV. Human papillomavirus prevalence among women with cervical intraepithelial neoplasia III and invasive cervical cancer from Goiania, Brazil. Mem Inst Oswaldo Cruz. 2003;98(2):181-4. http://doi. org/10.1590/S0074-02762003000200003
- Tozetti IA, Scapulatempo IDL, Kawski VL, Ferreira AW, Levi JE. Multiple types of human papillomavirus in cervical samples in women in Campo Grande, MS, Brazil. Braz J Infect Dis. 2006;10(5):309-10. http:// dx.doi.org/10.1590/S1413-86702006000500001
- 36. Castilho JL, Levi JE, Luz PM, Cambou MC, Vanni T, de Andrade A, et al. A cross-sectional study of high-risk human papillomavirus clustering and cervical outcomes in HIV-infected women in Rio de Janeiro, Brazil. BMC Cancer. 2015;15:478. http://dx.doi.org/10.1186/s12885-015-1486-4
- Pitta DR, Campos EA, Sarian LO, Rovella MS, Derchain SFM. Prevalência dos HPV 16, 18, 45 e 31 em mulheres com lesão cervical. Rev Bras Ginecol Obstet. 2010;32(7):315-20. http://dx.doi.org/10.1590/S0100-72032010000700002

- Giraldo PC, Silva MJPMA, Fedrizzi EN, Gonçalves AKS, Amaral RLG, Eleutério Junior J, et al. Prevenção da infecção por HPV e lesões associadas com o uso de vacinas. DST – J Bras Doenças Sex Transm. 2008;20(2):132-40.
- Gardasil 9. Summary of Product Characteristics [Internet]. European Medicines Agency. London, UK: European Medicines Agency; 2016 [cited on Mar. 8, 2017]. Available at: http://www.ema.europa.eu/docs/ en\_GB/document\_library/EPAR\_-\_Product\_Information/human/003852/ WC500189111.pdf
- Schauner S, Lyon C. Bivalent HPV Recombinant Vaccine (Cervarix) for the Prevention of Cervical Cancer. Am Fam Physician. 2010;82(12):1541-2.
- Borsatto AZ, Vidal MLB, Rocha RCNP. Vacina contra o HPV e a prevenção do câncer do colo do útero: subsídios para a prática. Rev Bras Cancerol. 2011;57(1):67-74.
- Santana ACC, Schmitz K. As vacinas profiláticas contra o HPV: uma revisão da literatura [essay]. Joinville: Católica de Santa Catarina; 2015.
- 43. Brasil. Ministério da Saúde. Secretaria de Vigilância em Saúde Departamento de Vigilância das Doenças Transmissíveis Coordenação-Geral do Programa Nacional de Imunizações. Informe técnico da ampliação da oferta das vacinas papilomavírus humano 6, 11, 16 e 18 (recombinante) – vacina HPV quadrivalente e meningocócica C (conjugada). Brasil: Ministério da Saúde; 2018 [cited on Jul. 31, 2019]. Available at http://portalarquivos2.saude.gov. br/images/pdf/2018/marco/14/Informe-T--cnico-HPV-MENINGITE.pdf
- Zhai L, Tumban E. Gardasil-9: A global survey of projected efficacy. Antiviral Res. 2016;130:101-9. https://doi.org/10.1016/j. antiviral.2016.03.016
- Cuzick J. Gardasil 9 joins the fight against cervix cancer. Expert Rev Vaccines. 2015;14(8):1047-9. https://doi.org/10.1586/14760584.2015.1051470
- Zardo GP, Farah FP, Mendes FG, Franco CAGS, Molina GVM, Melo GN, et al. Vacina como agente de imunização contra o HPV. Ciênc Saúde Coletiva. 2014;19(9):3799-808. http://dx.doi.org/10.1590/1413-81232014199.01532013
- Agência Nacional de Vigilância Sanitária. Registrada vacina do HPV contra 9 subtipos do vírus [Internet]. Agência Nacional de Vigilância Sanitária; 2017 [cited on Jul. 31, 2019]. Available at: http://portal.anvisa. gov.br/noticias/-/asset\_publisher/FXrpx9qY7FbU/content/registradavacina-do-hpv-contra-9-subtipos-do-virus/219201

- Merck Vaccines. Gardasil.9 Human Papillomavirus 9-valent Vaccine, Recombinant. Eficácia do GARDASIL 9 [Internet]. VACC-1246170-0001 10/18 [cited on Jul. 31, 2019]. Available at: https://www.merckvaccines.com/ Products/Gardasil9/efficacy#add4Types Acesso em: 31 de jul. de 2019.
- 49. Brasil. Ministério da Saúde. Secretaria de Vigilância em Saúde. Departamento de Vigilância das Doenças Transmissíveis. Coordenação-Geral do Programa Nacional de Imunizações. Informe técnico sobre a vacina Papilomavírus Humano (HPV) na atenção básica [Internet]. Brasil: Ministério da Saúde; 2018 [cited on Jul. 31, 2019]. Available at: http:// portalarquivos2.saude.gov.br/images/pdf/2015/junho/26/Informe-T-cnico-Introdu----o-vacina-HPV-18-2-2014.pdf
- IARC. Monographs on the evaluation of carcinogenic risks to humans. In: Cancer IAfRoCancer IAfRo, editor. HPV. Lyon: IARC; 2012.
- Jit M, Brisson M, Portnoy A, Hutubessy R. Cost-eectiveness of female human papillomavirus vaccination in 179 countries: a PRIME modelling study. Lancet Glob Health. 2014;2(7):e406-14. http://dx.doi.org/10.1016/ S2214-109X(14)70237-2
- Chesson HW, Meites E, Ekwueme DU, Saraiya M, Markowit LE. Costeffectiveness of nonavalent HPV vaccination among males aged 22 through 26 years in the United States. Vaccine. 2018;36(29):4362-8. https://doi.org/10.1016/j.vaccine.2018.04.071
- Wolff E, Elfström KM, Cange HH, Larssona S, Englund H, Sparén P, et al. Cost-effectiveness of sex-neutral HPV-vaccination in Sweden, accounting for herd-immunity and sexual behaviour. Vaccine. 2018;36(34):5160-5. https://doi.org/10.1016/j.vaccine.2018.07.018

### Address for correspondence:

### AISLAN CRISTINA RHEDER FAGUNDES PASCOAL

Rua Dr. Silvio Henrique Braune, 22 – Centro Nova Friburgo (RJ), Brazil CEP: 28625-650 E-mail: aislanfagundes@id.uff.br

Received on: 08.09.2019 Approved on: 09.10.2019





# ~ 2021